SECURITIES AND EXCHANGE COMMISSION

FORM 10-Q

Quarterly report pursuant to sections 13 or 15(d)

Filing Date: 2010-11-05 | Period of Report: 2010-09-30 SEC Accession No. 0000950123-10-101366

(HTML Version on secdatabase.com)

FILER

MOVE INC

CIK:1085770| IRS No.: 954438337 | State of Incorp.:DE | Fiscal Year End: 1231 Type: 10-Q | Act: 34 | File No.: 000-26659 | Film No.: 101166844 SIC: 6531 Real estate agents & managers (for others) Mailing Address 30700 RUSSELL RANCH RD WESTLAKE VILLAGE CA 91362

Business Address 30700 RUSSELL RANCH RD WESTLAKE VILLAGE CA 91362 805-557-2300

Copyright © 2012 www.secdatabase.com. All Rights Reserved. Please Consider the Environment Before Printing This Document

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

Form 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES $\mathbf{\nabla}$ **EXCHANGE ACT OF 1934**

For the quarterly period ended September 30, 2010

or

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from to

Commission File Number 000-26659

Move, Inc.

(Exact Name of Registrant as Specified in its Charter)

Delaware

(State or Other Jurisdiction of Incorporation or Organization)

910 East Hamilton Avenue **Campbell**, California (Address of Principal Executive Offices)

(805) 557-2300

(Registrant's Telephone Number, including Area Code)

(Former Name, Former Address and Former Fiscal Year, If Changed Since Last Report)

Indicate by check mark whether registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes ☑ No □

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes \Box No \Box

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer," and "smaller reporting company" in Rule 12b-2 of the Exchange Act

Large Accelerated Filer \Box Accelerated Filer ☑ Non-Accelerated Filer Smaller reporting company \Box (Do not check if a smaller reporting company)

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes \Box No \Box

At November 2, 2010, the registrant had 158,383,347 shares of its common stock outstanding.

Copyright © 2012 www.secdatabase.com. All Rights Reserved. Please Consider the Environment Before Printing This Document

95-4438337 (I.R.S. Employer Identification No.)

> 95008 (Zip Code)

Copyright © 2012 www.secdatabase.com. All Rights Reserved. Please Consider the Environment Before Printing This Document

INDEX

PART I – FINANCIAL INFORMATION

Page

Item 1.	Condensed Consolidated Financial Statements	3
	Condensed Consolidated Balance Sheets at September 30, 2010 (unaudited) and December 31, 2009	3
	Condensed Consolidated Statements of Operations for the Three and Nine Months ended September 30, 2010	4
	and 2009 (unaudited)	
	Condensed Consolidated Statements of Cash Flows for the Nine Months ended September 30, 2010 and 2009	5
	(unaudited)	
	Notes to Unaudited Condensed Consolidated Financial Statements	6
<u>Item 2.</u>	Management's Discussion and Analysis of Financial Condition and Results of Operations	15
<u>Item 3.</u>	Quantitative and Qualitative Disclosures About Market Risk	22
<u>Item 4.</u>	Controls and Procedures	22
	<u>PART II – OTHER INFORMATION</u>	
<u>Item 1.</u>	Legal Proceedings	22
Item 1A.	Risk Factors	22

<u>Item IA.</u>	<u>Risk Factors</u>	22
<u>Item 2.</u>	Unregistered Sales of Equity Securities and Use of Proceeds	23
<u>Item 3.</u>	Defaults Upon Senior Securities	23
<u>Item 4.</u>	[Removed and Reserved]	23
<u>Item 5.</u>	Other Information	23
<u>Item 6.</u>	Exhibits	23
SIGNATURES	5	24
EV 10 74		

EX-10.74 EX-31.1 EX-31.2 EX-32.1 EX-32.2

Move[®], REALTOR.com[®], Top Producer[®], and Moving.comTM are our trademarks or are exclusively licensed to us. This quarterly report on Form 10-Q contains trademarks of other companies and organizations. REALTOR[®] is a registered collective membership mark that may be used only by real estate professionals who are members of the National Association of REALTORS[®] and subscribe to its code of ethics.

2

PART I. FINANCIAL INFORMATION

Item 1. Condensed Consolidated Financial Statements

MOVE, INC.

CONDENSED CONSOLIDATED BALANCE SHEETS

	September 30, 2010	December 31, 2009
	(Unaudited)	
	(In tho	usands)
ASSETS		
Current assets:	Ф 155 (Q1	¢ 106 047
Cash and cash equivalents	\$ 155,621	\$ 106,847
Accounts receivable, net	9,814	10,782
Other current assets	9,856	12,101
Total current assets	175,291	129,730
Property and equipment, net	21,424	21,139
Long-term investments	-	111,800
Investment in unconsolidated joint venture	7,290	6,649
Goodwill, net	24,581	16,969
Intangible assets, net	8,672	3,460
Other assets	1,288	1,548
Total assets	\$ 238,546	\$ 291,295
Current liabilities:		
Accounts payable	\$ 4,352	\$ 5,545
Accrued expenses	18,964	18,335
Deferred revenue	15,581	15,951
Line of credit	-	64,630
Total current liabilities	38,897	104,461
Other non-current liabilities	3,527	1,096
Total liabilities	42,424	105,557
Commitments and contingencies (see note 15)		
Series B convertible preferred stock	115,565	111,541
Stockholders' equity:		
Series A convertible preferred stock	-	-
Common stock	158	156
Additional paid-in capital	2,122,745	2,112,613
Accumulated other comprehensive income (loss)	374	(17,116)
Accumulated deficit	(2,042,720)	(2,021,456)
Total stockholders' equity	80,557	74,197

\$ 238,546

\$ 291,295

The accompanying notes are an integral part of these unaudited Condensed Consolidated Financial Statements.

MOVE, INC.

CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS

	Three Mon Septem		Nine Mont Septem	
	2010	2009 (In thousands, except	2010 t per share amounts)	2009
Revenue	\$50,256	(Unau \$52,866	dited) \$148,590	\$162,371
Cost of revenue	10,766	12,014	32,782	37,465
Gross profit	39,490	40,852	115,808	124,906
01055 pront	59,490	40,032	115,000	124,900
Operating expenses:				
Sales and marketing	18,631	18,787	55,835	60,936
Product and web site development	8,855	7,650	25,517	20,458
General and administrative	10,877	16,226	32,366	51,227
Amortization of intangible assets	139	107	348	366
Litigation settlement	-	-	-	975
Restructuring charges (see note 4)		(1,192)		(1,192)
Total operating expenses	38,502	41,578	114,066	132,770
Operating income (loss) from continuing operations	988	(726)	1,742	(7,864)
Interest income, net	33	279	767	728
Earnings of unconsolidated joint venture	342	_	641	-
Impairment of auction rate securities	-	_	(19,559)	_
Other income (expense), net	(42)	1,250	(1,144)	1,741
Income (loss) from continuing operations before income taxes	1,321	803	(17,553)	(5,395)
Income tax (benefit) expense	(404)	50	(313)	227
Income (loss) from continuing operations	1,725	753	(17,240)	(5,622)
Loss from discontinued operations	-	(196)	(17,240)	(445)
Gain on disposition of discontinued operations	_	(1)0)	_	2,303
	1.725		(17.240)	
Net income (loss)	1,725	557	(17,240)	(3,764)
Convertible preferred stock dividend and related accretion	(1,350)	(1,315)	(4,024)	(3,920)
Net income (loss) applicable to common stockholders	\$375	<u>\$(758</u>)	<u>\$(21,264</u>)	<u>\$(7,684</u>)
Basic income (loss) per share applicable to common stockholders: (see note 10)				
Continuing operations	\$0.00	\$(0.00)	\$(0.14)	\$(0.06)
Discontinued operations	-	(0.00)	-	0.01
Basic net income (loss) per share applicable to common stockholders	\$0.00		$\frac{\$(0, 14,)}{\$(0, 14,)}$	
Basic net income (loss) per share appreable to common stockholders	\$0.00	<u>\$(0.00</u>)	<u>\$(0.14</u>)	<u>\$(0.05</u>)
Diluted income (loss) per share applicable to common stockholders: (see note 10)				
Continuing operations	\$0.00	\$(0.00)	\$(0.14)	\$(0.06)
Discontinued operations	-	(0.00)	-	0.01
Diluted net income (loss) per share applicable to common stockholders	\$0.00	\$(0.00)	\$(0.14)	\$(0.05)
Shares used to calculate basic and diluted net income (loss) per share applicable to common stockholders: (see note 10)				
Basic	155,711	153,344	154,957	153,139
Diluted	159,527	153,344	154,957	153,139

The accompanying notes are an integral part of these unaudited Condensed Consolidated Financial Statements.

MOVE, INC.

CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS

	Nine Mont Septem	
	2010	2009
	(In thou (Unau	,
Cash flows from operating activities:	(Unau	uncu)
Net loss	\$(17,240)	\$(3,764)
Adjustments to reconcile net loss to net cash provided by continuing operating activities:		
Loss from discontinued operations	-	445
Gain on disposition of discontinued operations	-	(2,303)
Depreciation	7,637	7,853
Amortization of intangible assets	348	366
Provision for doubtful accounts	(84)	1,024
Gain on sale of assets	-	(1,308)
Impairment of auction rate securities	19,559	—
Stock-based compensation and charges	5,765	15,647
Earnings of unconsolidated joint venture	(641)	_
Change in market value of embedded derivative liability	-	(536)
Other non-cash items	(153)	(83)
Changes in operating assets and liabilities, net of acquisitions and discontinued operations:		
Accounts receivable	1,346	130
Other assets	1,059	(998)
Accounts payable and accrued expenses	675	155
Deferred revenue	(602)	(6,433)
Net cash provided by continuing operating activities	17,669	10,195
Net cash used in discontinued operating activities	-	(1,843)
Net cash provided by operating activities	17,669	8,352
Cash flows from investing activities:	(7.020)	(7.711.)
Purchases of property and equipment	(7,838)	(7,711)
Acquisitions, net of cash acquired Proceeds from the sale of auction rate securities	(12,371)	-
	109,841	-
Principal payments on notes receivable	1,000	-
Proceeds from sale of assets	- 14	1,355
Proceeds from the sale of marketable equity securities	14	
Net cash provided by (used in) continuing investing activities	90,646	(6,356)
Net cash provided by discontinued investing activities	_	1,739
Net cash provided by (used in) investing activities	90,646	(4,617)
Cash flows from financing activities:		
Proceeds from exercise of stock options	4,467	1,859
Restricted cash	462	2,747
Gross proceeds from line of credit	64,700	_
Gross principal payments on line of credit	(129,330)	-
Proceeds from loan payable	316	-
Principal payments on loan payable	(58)	—
		(1.0(4))
	(98)	(1.064)
Tax payment related to net share settlements of restricted stock awards Payments on capital lease obligations	(98)	(1,064) (339)

Change in cash and cash equivalents	48,774	6,938
Cash and cash equivalents, beginning of period	106,847	108,935
Cash and cash equivalents, end of period	\$155,621	\$115,873

The accompanying notes are an integral part of these unaudited Condensed Consolidated Financial Statements.

MOVE, INC.

NOTES TO UNAUDITED CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

1. Business

Move, Inc. and its subsidiaries (the "Company") operate an online network of web sites for real estate search, finance, moving and home enthusiasts and provide a valuable resource for consumers seeking the online information and connections they need regarding real estate. The Company's flagship consumer web sites are Move.com, REALTOR.com® and Moving.com. The Company also provides lead management software for real estate agents through its Top Producer® business.

2. Principles of Consolidation and Basis of Presentation

The accompanying financial statements are consolidated and include the financial statements of Move, Inc. and its majority-owned subsidiaries. All significant intercompany balances and transactions have been eliminated in consolidation. The Company has evaluated all subsequent events through the date the financial statements were issued.

Investments in private entities where the Company holds no more than a 50% ownership interest and does not exercise control are accounted for using the equity method of accounting and the investment balance is included in investment in unconsolidated joint venture, while the Company's share of the investees' results of operations is included in earnings of unconsolidated joint venture.

The Company's unaudited Condensed Consolidated Financial Statements have been prepared in accordance with accounting principles generally accepted in the United States of America ("GAAP"), including those for interim financial information, and with the instructions for Form 10-Q and Article 10 of Regulation S-X issued by the Securities and Exchange Commission ("SEC"). Accordingly, they do not include all of the information and note disclosures required by GAAP for complete financial statements. These statements are unaudited and, in the opinion of management, all adjustments (which include only normal recurring adjustments) considered necessary for a fair presentation have been included. These unaudited Condensed Consolidated Financial Statements should be read in conjunction with the audited financial statements and notes thereto included in the Company's Form 10-K for the year ended December 31, 2009, which was filed with the SEC on March 5, 2010. The results of operations for these interim periods are not necessarily indicative of the operating results for a full year.

3. Acquisitions and Disposals

On September 20, 2010, the Company acquired all of the outstanding shares of Threewide Corporation ("Threewide") for approximately \$13.1 million in cash. Threewide is the operator of ListHub, an online real estate listing syndicator and provider of advanced performance reporting solutions for brokers, real estate franchises, multiple listing services and real estate web sites. The total purchase price has been allocated to the assets acquired, including intangible assets, and liabilities assumed, based on their respective fair values. The \$13.1 million purchase price was allocated \$0.6 million to net tangible assets (which included \$0.7 million of cash acquired), \$5.1 million to intangible assets with estimated useful lives of five years, \$0.5 million to indefinite lived trade name and trademarks, and the remaining \$6.9 million was allocated to goodwill. In connection with the purchase accounting, the Company recorded a net deferred tax liability of \$0.2 million associated with the indefinite lived intangible and an income tax benefit of \$0.5 million (see Note 13), resulting in additional goodwill of \$0.7 million being recorded. At September 30, 2010, the Company had goodwill of \$7.6 million and net intangible assets of \$5.5 million associated with the Threewide acquisition. The financial results of Threewide are included in the Company's Condensed Consolidated Financial Statements from the date of acquisition. Pro forma financial information for this acquisition has not been presented because the effects were not material to the Company's historical consolidated financial statements.

In the second quarter of 2008, the Company decided to divest its Welcome Wagon business. On June 22, 2009, the Company closed the sale of the business for a sales price of \$2.0 million. The Company received \$1.0 million in cash and a \$1.0 million promissory note. The principal balance of the note, which was originally due on or before October 1, 2010, was paid in full in July 2010. The outstanding principal bore an interest rate of 7% per annum, with quarterly interest payments due commencing on October 1, 2009. The transaction resulted in a gain on disposition of discontinued operations of \$1.2 million for the nine months ended September 30, 2009.

As part of the sale in 2002 of the Company's ConsumerInfo division to Experian Holding, Inc. ("Experian"), \$10.0 million of the purchase price was put in escrow to secure the Company's indemnification obligations (the "Indemnity Escrow"). Under the terms of the stock purchase agreement, the Company's maximum potential liability for claims by Experian was capped at \$29.3 million less the balance in the Indemnity Escrow, which amount was approximately \$8.5 million on March 31,

2009. During 2008, Experian demanded \$29.3 million in indemnity payments. The Company denied liability and a bifurcated arbitration proceeding ensued to resolve the dispute. Subsequent to the completion of the first phase of the arbitration proceedings, on April 20, 2009, the parties settled the dispute and entered into a full release of all claims under which Experian received \$7.4 million from the Indemnity Escrow and the Company received the balance of the escrow of \$1.1 million, which is included in gain on disposition of discontinued operations for the nine months ended September 30, 2009.

Pursuant to ASC 205-20-45 "Discontinued Operations", the Company's Consolidated Financial Statements for all periods presented reflects the reclassification of its Welcome Wagon business as discontinued operations. Accordingly, the revenue, costs and expenses, and cash flows of this business have been excluded from the respective captions in the Condensed Consolidated Statements of Operations and Condensed Consolidated Statements of Cash Flows and have been reported as "Income (loss) from discontinued operations," net of applicable income taxes of zero; and as "Net cash used in discontinued operating activities." Total revenue and income (loss) from discontinued operations are reflected below (in thousands):

		For the Three Months Ended September 30,		e Months Ended mber 30,
	2010	2009	2010	2009
Revenue	\$-	\$-	\$ -	\$9,609
Total operating expenses	_	196	-	9,009
Restructuring charges	-	-	-	1,045
Loss from discontinued operations	\$-	\$(196)	\$ -	\$(445)
Gain on disposition of discontinued operations	\$-	\$-	\$-	\$2,303

In the first quarter of 2009, the Company incurred a restructuring charge from discontinued operations of \$1.1 million associated with lease termination charges and employee termination costs. There are no additional amounts to be paid as part of the restructuring charge as of September 30, 2010.

In July 2009, the Company sold certain product lines associated with its Enterprise business for a sale price of approximately \$1.4 million in cash. The transaction resulted in a gain on sale of assets of \$1.3 million which is reflected in other income, net in the Company's Condensed Consolidated Statements of Operations for the three and nine months ended September 30, 2009.

4. Restructuring Charges

In the third and fourth quarters of 2008, the Company's Board of Directors approved restructuring and integration plans with the objective of eliminating duplicate resources and redundancies and implementing a new operating structure to lower total operating expenses. As a result of these plans, the Company incurred a restructuring charge from continuing operations of \$4.4 million for the year ended December 31, 2008. Included in this charge were lease obligations and related charges of \$3.0 million for the consolidation of the Company's operations in Westlake Village, California and the vacancy of a portion of the leased facility. On September 3, 2009, the Company entered into a new lease agreement for its Westlake Village facility. Under the terms of the lease, the Company was leasing only a portion of the facility but was continuing to occupy its current space in that facility until construction was completed on the new space. The Company's obligations under the old lease were terminated and, as a result, the remaining restructuring reserve was reversed, resulting in a \$1.2 million credit to restructuring charges for the three and nine months ended September 30, 2009.

5. Long-term Investments

The following table summarizes the Company's long-term investments (in thousands):

	September 30, 2010			December 31, 2009			
		Net			Net		
	Adjusted Cost	Realized Loss	Carrying Value	Adjusted Cost	Unrealized Loss	Carrying Value	
Long-term investments							
Corporate auction rate securities	\$-	\$-	\$-	\$129,400	\$(17,600)	\$111,800	
Total long-term investments	\$-	\$-	\$-	\$129,400	\$(17,600)	\$111,800	

The Company's long-term investments as of December 31, 2009 consisted of high-grade (primarily AAA rated) student loan auction rate securities issued by student loan funding organizations, which loans are 97% guaranteed under FFELP (Federal Family Education Loan Program). These auction rate securities ("ARS") were intended to provide liquidity via an auction process that reset the interest rate, generally every 28 days, allowing investors to either roll over their holdings or sell them at par. In February 2008, auctions for the Company's investments in these securities failed to settle on their respective settlement dates. Consequently, the investments were not liquid and the Company was not going to be able to access these funds until a future auction of these investments was successful, the securities matured or a

buyer was found outside of the auction process. Maturity dates for these ARS investments ranged from years 2030 to 2047 with principal distributions

occurring on certain securities prior to maturity. As of December 31, 2009, the Company classified \$111.8 million of the ARS investment balance as Long-term Investments because of the Company's inability to determine when these investments would become liquid.

As of December 31, 2009, the Company had recorded a temporary loss related to the ARS of \$17.6 million that was included in Other Comprehensive Income on the Company's Condensed Consolidated Balance Sheet. At a board meeting on March 24, 2010, the Board of Directors and Management discussed the recent passage of the Health Care Reform Bill that contained a provision eliminating FFELP, a significant change in student loan funding. In management's opinion, this change, along with other market factors, created additional uncertainty in the student loan auction rate securities market. As a result, the Board of Directors and Management changed its intent, which had been to hold these securities, and decided to sell the entire portfolio of ARS and, thereafter, the Company began to actively market the sale to third parties. The Company reviews its potential investment impairments in accordance with ASC 320 "Investment – Debt and Equity Securities" and the related guidance issued by the FASB and SEC in order to determine the classification of the impairment as "temporary" or "other-than-temporary." A temporary impairment charge results in an unrealized loss being recorded in the other comprehensive income (loss) component of stockholder's equity. An other-than-temporary impairment charge is recorded as a realized loss in the Condensed Consolidated Statement of Operations and reduces net income (loss) for the applicable accounting period. The differentiating factors between temporary and other-than-temporary impairment are primarily the length of the time and the extent to which the market value has been less than cost, the financial condition and near-term prospects of the issuer, and the ability and intent of the holder to hold the investment until maturity or its value recovers. Prior to March 24, 2010, the Company had not intended to sell, nor was it not more likely than not that the Company would be required to sell before the recovery of its amortized cost basis and, as such, the loss was considered temporary. On March 24, 2010, as indicated above, the Company changed its intent to hold the ARS and, therefore, the impairment was reclassified to an other-than-temporary loss.

In April 2010, the Company completed a sale of the entire portfolio of ARS for \$109.8 million (par value of \$129.4 million) to a broker in a secondary market. As a result of the sale, an other-than-temporary loss of \$19.6 million was recorded as Impairment of Auction Rate Securities in the Company's Condensed Consolidated Statement of Operations for the nine months ended September 30, 2010. The transaction costs of approximately \$1.0 million associated with this transaction were recorded as other expense for the nine months ended September 30, 2010.

6. Fair Value Measurements

Financial assets and liabilities included in the Company's financial statements and measured at fair value on a recurring basis as of September 30, 2010 and December 31, 2009 are classified based on the fair value hierarchy in the table below (in thousands):

	Fair Value Measurement							
		September	30, 2010			December	· 31, 2009	
	Total	Level 1	Level 2	Level 3	Total	Level 1	Level 2	Level 3
Description:								
Assets:								
Cash and cash equivalents (1)	\$155,621	\$155,621	\$-	\$-	\$106,847	\$106,847	\$-	\$ -
Long-term investments (2)	_	_	_	-	111,800	_	_	111,800
Total assets at fair value	\$155,621	\$155,621	<u>\$-</u>	<u>\$</u> -	\$218,647	\$106,847	\$ -	\$111,800

(1) Cash and cash equivalents consist primarily of treasury bills with original maturity dates of three months or less and money market funds for which the Company determines fair value through quoted market prices.

Long-term investments consisted of student loan, FFELP-backed, ARS issued by student loan funding organizations. Prior to March 31, 2010, the Company used a discounted cash flow model to determine the estimated fair value of its investment in ARS. The assumptions

(2) used in preparing the discounted cash flow model included estimates for interest rates, timing and amount of cash flows and the Company's expected holding period of the ARS. As discussed in Note 5, the Company completed the sale of its entire portfolio of ARS in April 2010.

The following table provides a reconciliation of the beginning and ending balances for the major class of assets and liabilities measured at fair value using significant unobservable inputs (Level 3) (in thousands):

	Long-term
	Investments
Balance at January 1, 2010	\$ 111,800
Transfers out of Level 3	(109,841)
Total losses included in earnings	(19,559)
Total gains included in other comprehensive income	17,600
Purchases, sales, issuances and settlements, net	-
Balance at September 30, 2010	\$ –

Certain assets and liabilities are measured at fair value on a non-recurring basis. That is, the assets and liabilities are not measured at fair value on an ongoing basis, but are subject to fair value adjustments in certain circumstances (e.g. when there is evidence of impairment). At September 30, 2010, the Company had no significant non-financial assets or liabilities that had been adjusted to fair value subsequent to initial recognition.

7. Revolving Line of Credit

The Company had a revolving line of credit with a major financial institution, providing for borrowings of up to \$64.7 million. The line of credit was fully paid in April 2010 and expired in May 2010.

8. Goodwill and Intangible Assets

Goodwill increased \$7.6 million from \$17.0 million as of December 31, 2009 to \$24.6 million as of September 30, 2010 due to the acquisition of Threewide as described in Note 3.

The Company has both indefinite-lived and definite-lived intangibles. Indefinite-lived intangibles consist of \$2.5 million of trade names and trademarks. Definite-lived intangible assets consist of certain trade names, trademarks, brand names, content syndication agreements, purchased technology, customer contracts and related customer relationships, non-contractual customer relationships, and other miscellaneous agreements. The definite-lived intangibles are amortized over the expected period of benefit. There are no expected residual values related to these intangible assets. Intangible assets, by category, are as follows (in thousands):

	September 30, 2010		Decem	ber 31, 2009
	Gross Amount	Accumulated Amortization	Gross Amount	Accumulated Amortization
Trade names, trademarks, brand names, and domain names	\$3,060	\$ 517	\$2,530	\$ 516
Content syndication agreements	3,800	21	—	-
Purchased technology	1,400	917	1,400	767
National Association of Realtors ("NAR") operating agreement	1,578	1,315	1,578	1,202
Other	2,680	1,076	1,450	1,013
Total	\$12,518	\$ 3,846	\$6,958	\$ 3,498

Amortization expense for intangible assets was \$0.1 million and \$0.3 million for each of the three and nine month periods ended September 30, 2010 and 2009, respectively.

Amortization expense for the next five years is estimated to be as follows (in thousands):

Years Ended December 31,	Amount
2010 (remaining 3 months)	\$355
2011	1,421
2012	1,346
2013	1,104
2014	1,072

9. Stock-Based Compensation and Charges

The Company accounts for stock issued to non-employees in accordance with the provisions of ASC 505-50 "Equity-Based Payments to Non-Employees" (formerly SFAS No. 123 and EITF No. 96-18).

The Company grants restricted stock awards to non-employee members of its Board of Directors as compensation (except any director who is entitled to a seat on the Board of Directors on a contractual basis). These shares, subject to certain terms

Copyright © 2012 www.secdatabase.com. All Rights Reserved. Please Consider the Environment Before Printing This Document

and restrictions, will vest on the third anniversary of their issuance and the costs are being recognized over their respective vesting periods. During the nine months ended September 30, 2010, the Company granted 139,098 shares of restricted stock to the non-employee members of its Board of Directors. During the nine months ended September 30, 2009, the Company granted 60,000 shares of restricted stock to the members of the ad hoc Executive Committee of its Board of Directors. Half of these shares vested on the grant date and half of the shares vested one year from the grant date. Additionally, during the nine months ended September 30, 2009, the Company granted 175,420 shares of restricted stock to all non-employee members of its Board of Directors. There were 432,105 and 453,713 unvested shares of restricted stock issued to members of the Company's Board of Directors as of September 30, 2010 and 2009, respectively. Total cost recognized was \$0.1 million for the three months ended September 30, 2010 and 2009, and \$0.3 million and \$0.4 million for the nine months ended September 30, 2010 and 2009, respectively, and is included in stock-based compensation and charges.

Restricted Stock

During the nine months ended September 30, 2009, the Company issued 1,800,000 shares of restricted stock to its new Chief Executive Officer ("CEO") as part of his employment agreement with the Company. These shares had a grant date fair value of \$2.7 million, 700,000 of which shares vested immediately, 500,000 of which shares vested one year from the grant date and 600,000 of which shares vest two years from the grant date, subject to certain terms and restrictions. The fair value of the 700,000 immediately vested shares was recognized as stock-based compensation immediately, with the fair value of the remaining shares being amortized over their respective vesting periods. During the nine months ended September 30, 2009, the CEO returned 700,000 shares of common stock, with a fair value of \$1.1 million, to reimburse the Company for his share of income tax withholdings due as a result of this transaction. The \$1.1 million payment to the relevant taxing authorities is reflected as a financing activity within the Condensed Consolidated Statement of Cash Flows for the nine months ended September 30, 2009. Total cost recognized during the three months ended September 30, 2010 and 2009 was \$0.1 million and \$0.3 million, respectively, and \$0.4 million and \$1.9 million for the nine months ended September 30, 2010 and 2009, respectively, and is included in stock-based compensation and charges.

During the year ended December 31, 2009, the Company issued 350,000 shares of restricted stock to two other executives newly hired by the Company, as part of their employment agreements with the Company. These shares had an aggregate grant date fair value of \$0.9 million. These shares vest in three annual installments over the three year period following their respective grant dates, subject to certain terms and restrictions. Total cost recognized during the three and nine months ended September 30, 2010 was \$0.1 million and \$0.4 million, respectively, and \$0.1 million for the three and nine months ended September 30, 2009 and is included in stock-based compensation and charges. The Company made no grants of restricted stock to any of its executive officers or employees during the three and nine months ended September 30, 2010.

Performance Based Restricted Stock Units

During the nine months ended September 30, 2009, the Board of Directors awarded 700,000 shares of performance-based restricted stock units to its new Chief Executive Officer as part of his employment agreement with the Company. These awards will be earned based on the attainment of certain performance goals relating to the Company's revenues and EBITDA for the fiscal year ending December 31, 2011. The performance goals were established on April 9, 2010 and the shares had an aggregate grant date fair value of \$1.5 million. The implied service period is for the fiscal year ending December 31, 2011; therefore, there was no recognition of compensation expense for these units during the three and nine months ended September 30, 2010.

During the year ended December 31, 2009, the Board of Directors awarded 375,000 shares of performance-based restricted stock units to two other executives newly hired by the Company, as part of their employment agreements. These awards will be earned based on the attainment of certain performance goals relating to the Company's revenues and EBITDA for the fiscal years ending December 31, 2010, 2011 and 2012. The performance goals for fiscal year ending December 31, 2010 were established on March 29, 2010. The shares associated with the fiscal year ending December 31, 2010 had an aggregate grant date fair value of \$0.3 million. Total cost recognized for these units during the three and nine months ended September 30, 2010 was \$0.1 million and \$0.2 million, respectively, and is included in stock-based compensation. The performance goals for fiscal year ending December 31, 2011 were established on April 9, 2010. The shares associated with the fiscal year ending December 31, 2011 had an aggregate grant date fair value of \$0.3 million. Total cost recognized for these units during the three and nine months ended September 30, 2010 was \$0.1 million and \$0.2 million. Total cost recognized in stock-based compensation. The performance goals for fiscal year ending December 31, 2011 were established on April 9, 2010. The shares associated with the fiscal year ending December 31, 2011 had an aggregate grant date fair value of \$0.3 million. The implied service period is for the fiscal year ending December 31, 2011, therefore; there was no recognition of compensation expense for these units during the three and nine months ended September 30, 2010.

The Company awarded no new performance-based restricted stock units to any of its executive officers or employees during the nine months ended September 30, 2010.

Option Awards

The fair value of stock option awards is estimated on the date of grant using a Black-Scholes option valuation model that uses the ranges of assumptions in the following table. Our computation of expected volatility is based on a combination of historical and market-based implied volatility. The expected term is based on the Company's weighted average vesting period combined with the post-vesting holding period. The risk-free interest rates are based on U.S. Treasury zero-coupon bonds for the periods in which the options were granted.

Copyright © 2012 www.secdatabase.com. All Rights Reserved. Please Consider the Environment Before Printing This Document

	ſ	Three Months Ended September 30,		Nine Months Ended September 30,				
	2010		2009		2010		2009)
Risk-free interest rates	1.27%-	1.70%	0.19%-	2.53%	1.27%-2.	43%	0.11%	2.54%
Expected term (in years)	5.85		5.85		5.85		5.85	
Dividend yield	0	%	0	%	0	%	0	%
Expected volatility	80	%	85	%	80%-85	%	85	%

During the nine months ended September 30, 2009, the Company granted options to purchase 3,000,000 shares of the Company's common stock to its new Chief Executive Officer as part of his employment agreement with the Company. The grant date fair value of these options was \$3.2 million. 750,000 of such options were immediately vested, with the remaining options vesting ratably on a monthly basis over a period of three years beginning on the first anniversary of the grant date. As a result of the immediate vesting, the Company recorded additional stock-based compensation of \$0.7 million for the nine months ended September 30, 2009.

During the nine months ended September 30, 2009, the Company accelerated the vesting of previously-awarded stock options for two former executive officers and extended the time to exercise certain of these options for one of the former officers as part of their separation agreements. As a result, the Company recorded additional stock-based compensation expense of \$7.2 million for the nine months ended September 30, 2009. There were no such accelerations for the three and nine months ended September 30, 2010.

The following chart summarizes the stock-based compensation and charges that have been included in the following captions for each of the periods presented (in thousands):

		Three Months Ended September 30,		nths Ended mber 30,
	2010	2009	2010	2009
Cost of revenue	\$44	\$54	\$136	\$137
Sales and marketing	400	472	1,240	1,349
Product and web site development	453	166	1,243	493
General and administrative	962	3,030	3,146	13,668
Total from continuing operations	1,859	3,722	5,765	15,647
Total from discontinued operations	-	1	-	64
Total stock-based compensation and charges	\$1,859	\$3,723	\$5,765	\$15,711

10. Net Income (Loss) Per Share

The following table sets forth the computation of basic and diluted net income (loss) per share applicable to common stockholders for the periods indicated (in thousands, except per share amounts):

	Three Months Ended September 30,		Nine Months Ended September 30,	
	2010	2009	2010	2009
Numerator:				
Income (loss) from continuing operations	\$1,725	\$753	\$(17,240)	\$(5,622)
Income (loss) from discontinued operations	-	(196)	-	1,858
Net income (loss)	1,725	557	(17,240)	(3,764)
Convertible preferred stock dividend and related accretion	(1,350)	(1,315)	(4,024)	(3,920)
Net income (loss) applicable to common stockholders	\$375	\$(758)	\$(21,264)	\$(7,684)
Net income (loss) applicable to common stockholders from continuing operations	\$375	\$(562)	\$(21,264)	\$(9,542)
Net income (loss) applicable to common stockholders from discontinued operations	-	(196)	_	1,858
Net income (loss) applicable to common stockholders	\$375	<u>\$(758</u>)	\$(21,264)	\$(7,684)
Denominator:				
Basic weighted average shares outstanding	155,711	153,344	154,957	153,139
Add: dilutive effect of options and restricted stock	3,816	-	-	_
Fully diluted weighted average shares outstanding	159,527	153,344	154,957	153,139

	Three Months Ended September 30,		Nine Mont Septeml	
	2010	2009	2010	2009
Basic income (loss) per share applicable to common stockholders:				
Continuing operations	\$0.00	\$(0.00)	\$(0.14)	\$(0.06)
Discontinued operations	-	(0.00)	-	0.01
Net income (loss)	\$0.00	<u>\$(0.00</u>)	<u>\$(0.14</u>)	<u>\$(0.05</u>)
Diluted income (loss) per share applicable to common stockholders:				
Continuing operations	\$0.00	\$(0.00)	\$(0.14)	\$(0.06)
Discontinued operations	-	(0.00)	-	0.01
Net income (loss)	\$0.00	\$(0.00)	\$(0.14)	\$(0.05)

Because their effects would be anti-dilutive for the periods presented, the denominator in the above computation of diluted loss per share excludes preferred stock, stock options and warrants of 45,179,291 and 62,383,313 for the three and nine months ended September 30, 2010, respectively, and 66,094,890 and for the three and nine months ended September 30, 2009, respectively.

11. Other Comprehensive Income (Loss)

The components of other comprehensive income (loss) are (in thousands):

		Three Months Ended September 30,		hs Ended ber 30,
	2010	2009	2010	2009
Net income (loss)	\$1,725	\$557	\$(17,240)	\$(3,764)
Unrealized gain (loss) on marketable securities	-	—	(3)	(3)
Reclass of unrealized loss on auction rate securities	-	_	17,600	-
Foreign currency translation	3	83	(107)	93
Other comprehensive income (loss)	\$1,728	\$640	\$250	\$(3,674)

12. Related-party Transactions

The Company provided product development services to the National Association of Realtors ("NAR") and recognized \$0.4 million and \$2.4 million in revenues for the three and nine months ended September 30, 2009, respectively. The Company did not provide product development services to the NAR during the three and nine months ended September 30, 2010. The Company also makes payments to the NAR required under its operating agreement with the NAR and under certain other advertising agreements. Total amounts paid under these agreements were \$0.4 million and \$1.4 million for the three and nine months ended September 30, 2010, respectively, and \$0.5 million and \$1.3 million for the three and nine months ended September 30, 2009, respectively.

13. Income Taxes

As a result of historical net operating losses, the Company currently provides a full valuation allowance against its net deferred tax assets. The Company recorded an income tax benefit of \$0.4 million and \$0.3 million in the three and nine months ended September 30, 2010, respectively. The Company recorded income tax expense of \$0.1 million and \$0.2 million in the three and nine months ended September 30, 2009, respectively. In the three and nine months ended September 30, 2010, the Company recorded an income tax benefit of \$0.5 million as a result of a change in the valuation allowances resulting from the deferred tax liability established for the amortizable intangible assets acquired as part of the business combination described in Note 3. This income tax benefit was partially offset by \$0.1 million and \$0.2 million of state income tax expenses and a deferred tax provision related to amortization of certain indefinite lived intangible assets in the three and nine months ended state income taxes and a deferred tax provision related to amortization of certain indefinite lived intangible assets.

As of September 30, 2010, the Company does not have any accrued interest or penalties related to uncertain tax positions. The Company's policy is to recognize interest and penalties related to uncertain tax positions in income tax expense. The Company does not have any interest or penalties related to uncertain tax positions in income tax expense for the three and nine months ended September 30, 2010 and 2009. The tax years 1993-2009 remain open to examination by the major taxing jurisdictions to which the Company is subject.

14. Settlement of Disputes and Litigation

On November 12, 2008, Patricia Ramirez, on behalf of herself and all other similarly situated California account executives, filed a purported class action lawsuit in the Los Angeles Superior Court against Move, Inc., and its subsidiary

Move Sales, Inc., asserting failure to fully reimburse business expenses, unlawful wage deductions, failure to timely pay wages due at termination, failure to timely furnish accurate itemized wage statements, unfair business practices and declaratory relief. Subsequent to December 31, 2009, the Company and plaintiff's attorneys agreed to a tentative settlement of all claims brought by Ramirez on behalf of herself and all others in the purported class action. The amount of the settlement was accrued as of December 31, 2009 and was recorded in the Consolidated Statements of Operations for the year ended December 31, 2009. Such proposed settlement required final court approval. On August 24, 2010 the court entered final approval of the settlement. The settlement did not have a material effect on the Company's results of operations or cash flows for the year ended December 31, 2009.

15. Commitments and Contingencies

Legal Proceedings

The Company is currently involved in certain legal proceedings, as discussed in Note 23, "Commitments and Contingencies–Legal Proceedings," to our Consolidated Financial Statements contained in Item 8 in our Annual Report on Form 10-K for the year ended December 31, 2009 ("Annual Report") and below in this Note 15. As of the date of this Form 10-Q, and except as disclosed below, there have been no material developments in the legal proceedings disclosed in our Annual Report, and the Company is not a party to any other litigation or administrative proceedings that management believes will have a material adverse effect on the Company's business, results of operations, financial condition or cash flows.

In March 2010, Smarter Agent, LLC ("Smarter Agent") filed suit against Move, Inc., against our affiliate, RealSelect, Inc. ("RealSelect"), and also against other co-defendants Boopsie, Inc., Classified Ventures, LLC, Hotpads, Inc., IDX, Inc., Multifamily Technology Solutions, Inc. D/B/A MyNewPlace, Primedia, Inc., Consumer Source, Inc., Trsoft, Inc. D/B/A PlanetRE, Trulia, Inc., Zillow, Inc., and ZipRealty, Inc. in the United States District Court for the District of Delaware. The complaint alleges that the Company and RealSelect, Inc. infringe U.S. Patents 6,385,541; 6,496,776; and 7,072,665 ("Patents in Suit") by offering an iPhone application for the REALTOR.com® web site and requested an unspecified amount of damages (including enhanced damages for willful infringement and attorneys' fees) and an injunction. On August 31, 2010, co-defendants Boopsie, Inc., Classified Ventures, LLC, Hotpads, Inc., IDX, Inc., Multifamily Technology Solutions Inc., Primedia, Inc., Consumer Source, Inc., Trsoft, Inc., Trulia, Inc., Zillow, Inc., and ZipRealty, Inc., filed requests for inter partes reexamination of the Patents in Suit. On September 30, 2010, the Company filed an answer and counter claims on behalf of Move and RealSelect. On October 22, 2010, Smarter Agent filed its answer to such counter claims. The Company intends to vigorously defend all claims. At this time, however, the Company is unable to express an opinion on the outcome of this case.

In June 2010, BanxCorp filed a lawsuit against Move, Inc., Dow Jones & Company, Inc., The New York Times Company, CNBC Inc., Cable News Network, Inc., MSNBC Interactive News, LLD, Fox News Network, LLC, AOL Inc., and LendingTree LLC in the United States District Court for the District of New Jersey. The first amended complaint alleges antitrust violations pursuant to Section 1 of the Sherman Antitrust Act and the New Jersey Antitrust Act, including allegations the Company and other defendants formed a cartel with BankRate to gain market dominance and unfair advantage over BanxCorp and other independent competitors by, among other alleged activities, fixing prices and allocating customers and markets with other BankRate website operators. The first amended complaint seeks a declaratory judgment, injunctive relief and unspecified amount of damages, including treble damages and attorneys' fees and legal costs. On August 3, 2010, motions to dismiss the case were filed jointly by the Company and the co-defendants. The Company intends to vigorously defend all claims. At this time, however, the Company is unable to express an opinion on the outcome of this case.

On February 28, 2007, in a patent infringement action against a real estate agent, Diane Sarkisian, pending in the U.S. District Court for the Eastern District of Pennsylvania ("the Sarkisian case"), Real Estate Alliance, Limited ("REAL"), moved to certify two classes of defendants: subscribers and members of the multiple listing service of which Sarkisian was a member, and customers of the Company who had purchased enhanced listings from the Company. The U.S. District Court in the Sarkisian case denied REAL's motion to certify the classes on September 24, 2007. On March 25, 2008, the U.S. District Court in the Sarkisian case stayed that case, and denied without prejudice all pending motions, pending the U.S. District Court of California's determination in the Move California Action (see below) of whether the Company's web sites infringe the REAL patents.

On April 3, 2007, in response to REAL' s attempt to certify our customers as a class of defendants in the Sarkisian case, the Company filed a complaint in the U.S. District Court for the Central District of California ("District Court") against REAL, and its licensing agent ("the Move California Action") seeking a declaratory judgment that the Company does not infringe U.S. Patent Nos. 4,870,576 and 5,032,989 ("the REAL patents"), that the REAL patents are invalid and/or unenforceable, and alleging several business torts and unfair competition. On August 8, 2007, REAL denied the Company's allegations, and asserted counterclaims against the Company for infringement of the REAL patents seeking compensatory damages, punitive damages, treble damages, costs, expenses, reasonable attorneys' fees and pre- and post-judgment interest. On February 28, 2008, REAL filed a motion for leave to amend its counter-claims, and to include NAR and the National Association of Home Builders ("NAHB") as individual defendants, as well as various brokers including RE/Max International ("RE/Max"), agents, Multiple Listing Services ("MLS"), new home builders, rental property owners, and technology providers and indicated that it

intended to seek to certify certain defendant classes. On March 11, 2008, REAL filed a separate suit in the District Court ("the REAL California Action") alleging infringement of the REAL patents against the same defendants it sought to include in its proposed amended counter-claims in the Move California Action, and also indicated that it intended to seek to certify the same defendant classes. The Company is not named as a defendant in the REAL California Action; however, the Company is defending NAR, NAHB and RE/Max in the REAL California Action was transferred to the same judge in the REAL California Action and in September, 2008, the court coordinated both cases and issued an order dividing the issues into two phases. Phase 1 addresses issues of patent validity and enforceability, whether Move websites infringe, damages, and liability of Move, NAR and NAHB. Phase 2 will address REAL's infringement claims related to the websites owned or operated by the remaining defendants and whether those defendants infringe the Real patents by using the Move websites. The District Court has stayed Phase 2 pending resolution of the issues in Phase 1.

On November 25, 2009, the court entered its claim construction order in the Move California Action. On January 20, 2010, the Move California Action parties filed a joint stipulation of non-infringement, and requested the District Court enter judgment of non-infringement. The District Court entered a final judgment of non-infringement on January 27, 2010.

In July, 2010, REAL filed its brief appealing the District Court's claim construction with the Federal Circuit Court of Appeals ("Circuit Court"), and in October, 2010, the Company filed its opposition. The Circuit Court has not set a date for oral argument. If the Circuit Court overturns all or part of the claim construction, the judgment would be vacated and the Move California Action would be remanded to the District Court for further litigation. If the Circuit Court upholds the District Court's claim construction, the Move California Action would be dismissed with prejudice. At this time, however, the Company is unable to express an opinion on the outcome of these cases.

Contingencies

From time to time, the Company is party to various other litigation and administrative proceedings relating to claims arising from its operations in the ordinary course of business. As of the date of this Form 10-Q and except as set forth herein, or in our Annual Report, the Company is not a party to any other litigation or administrative proceedings that management believes would have a material adverse effect on the Company's business, results of operations, financial condition or cash flows.

16. Supplemental Cash Flow Information

During the nine month period ended September 30, 2010:

The Company paid \$0.1 million in interest.

The Company received a trade-in allowance on the purchase of property and equipment of \$0.2 million.

The Company issued 139,098 share of restricted common stock to the non-employee members of its Board of Directors which vest over three years. The charge associated with these shares was \$0.3 million and is being recognized over the three-year vesting period.

The Company issued \$3.1 million in additional Series B Preferred Stock as in-kind dividends.

During the nine month period ended September 30, 2009:

The Company paid \$1.3 million in interest.

The Company issued 1,800,000 shares of restricted common stock to its new Chief Executive Officer with 700,000 shares vesting immediately, and, subject to certain terms and restrictions, 500,000 shares vesting one year from the grant date and 600,000 shares vesting two years from the grant date. The charge associated with these shares was \$2.7 million and is being recognized over the vesting periods.

The Company issued 60,000 shares of restricted common stock to the members of the ad hoc Executive Committee of its Board of Directors (except any director who is entitled to a seat on the Board of Directors on a contractual basis). Half of these shares vested on the grant date and half of the shares will vest one year from the grant date. The charge associated with these shares was \$0.1 million and was recognized over the one-year vesting period.

The Company issued 175,420 shares of restricted common stock to the non-employee members of its Board of Directors which vest over three years. The charge associated with these shares was \$0.4 million and is being recognized over the three-year vesting period.

The Company received a \$1.0 million promissory note in conjunction with the sale of its Welcome Wagon division.

The Company issued \$2.9 million in additional Series B Preferred Stock as in-kind dividends.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

This Form 10-Q and the following "Management's Discussion and Analysis of Financial Condition and Results of Operations" include "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995. This Act provides a "safe harbor" for forward-looking statements to encourage companies to provide prospective information about themselves so long as they identify these statements as forward-looking and provide meaningful cautionary statements identifying important factors that could cause actual results to differ from the projected results. All statements other than statements of historical fact that we make in this Form 10-Q are forward-looking. In some cases, you can identify these statements by forward-looking words such as "estimates," "expects," "anticipates," "projects," "plans," "intends," "believes," "might," "will," "should," or the negative of these terms and other comparable terminology, although not all forwardlooking statements are so identified. In particular, the statements herein regarding industry prospects and our future consolidated results of operations or financial position are forward-looking statements. Forward-looking statements reflect our current expectations and are inherently uncertain. Our actual results may differ significantly from our expectations. Factors that could cause or contribute to such differences include those discussed below and elsewhere in this Form 10-Q, as well as those discussed in our Annual Report on Form 10-K for the year ended December 31, 2009, and in other documents we file with the Securities and Exchange Commission ("SEC"). This Form 10-Q should be read in conjunction with our Annual Report, including the factors described under the caption Part 1, Item 1A, "Risk Factors" on our Form 10-K for the year ended December 31, 2009.

Our Business

Move, Inc. and its subsidiaries ("Move", "we", "our" or "us") operate an online network of web sites for real estate search, finance, moving and home enthusiasts and provide a valuable resource for consumers seeking the online information and connections they need regarding real estate. Our flagship consumer web sites are Move.com, REALTOR.com® and Moving.com. We also provide lead management software for real estate agents through our Top Producer® business.

On our web sites, we display comprehensive real estate property content, with over four million resale, new home and rental listings, as well as extensive move-related information and tools. We hold a significant leadership position in terms of web traffic and minutes, attracting an average of 10.8 million consumers to our network per month and 223 million minutes per month for the nine months ended September 30, 2010 according to comScore Media Metrix, a substantial lead over the next leading real estate site. We also have significant relationships with the real estate industry, including content agreements with approximately 900 Multiple Listing Services ("MLS") across the country and exclusive relationships with the National Association of REALTORS® ("NAR") and the National Association of Home Builders ("NAHB").

Business Trends and Conditions

In recent years, our business has been, and we expect may continue to be, influenced by a number of macroeconomic, industry-wide and product-specific trends and conditions:

Market and economic conditions.

In recent years, the U.S. economy has experienced low interest rates, and volatility in the equities markets. Through 2005, housing starts remained strong, while the supply of apartment housing generally exceeded demand. For a number of years prior to 2007, owning a home became much more attainable for the average consumer due to the availability of flexible mortgage options, which required minimal down payments and provided low interest rates. During this period, home builders spent less on advertising, given the strong demand for new houses, and homeowners who were looking to sell a home only had to list it at a reasonable price in most areas of the U.S. to sell in 60 days or less. Conversely, demand for rental units declined and apartment owners did not spend as much money on advertising, as they have sought to achieve cost savings during the difficult market for rentals. These trends had an impact on our ability to grow our business.

Beginning in the second half of 2006, the market dynamics seemed to reverse. Interest rates rose and mortgage options began to decline. The housing market became saturated with new home inventory in many large metropolitan markets and the available inventory of resale homes began to climb as demand softened. The impact of the rise in interest rates caused demand for homes to decline in mid-2007. In the second half of 2007, the availability of mortgage financing became very sparse. The lack of liquidity coupled with increased supply of homes and declining prices had a significant impact on real estate professionals, our primary customers.

Throughout 2008, market conditions continued to decline and in late September 2008, the stock market declines negatively impacted the liquidity of the markets in general and have contributed to the decline in consumer spending. With the exception of very few markets, new home starts have stalled. Consumer confidence declined and while mortgage rates have appeared to decline, the credit standards are perceived to be the tightest they have been in years. The combination of these factors has had a negative impact on the demand for homes. These changing conditions resulted in fewer home purchases and forced many real estate professionals to reconsider their marketing spend.

In 2006, we saw many customers begin to shift their dollars from conventional offline channels, such as newspapers and real estate guides, to the Internet. We saw many agents and brokers move their spending online and many home builders increased their marketing spend to move existing inventory, even as they slowed their production and our business grew as a result. However, as the market declined in 2008, the advertising spend by many of the large agents and brokers slowed and some of the medium and smaller brokers and agents reduced expenses to remain in business. This caused us to experience a decline in revenue in 2008 and 2009.

2009 was difficult for the real estate market and it has not improved in 2010. Delinquencies are expected to continue to be double that of foreclosures, causing uncertainty in the price floor within various markets. This coupled with the fact that banks have significantly tightened their credit standards for mortgage loans will make home purchases in the upper end of the market that much more difficult. We believe these market conditions will continue to put pressure on spending by real estate professionals and brokers in the next year.

Evolution of Our Product and Service Offerings and Pricing Structures

We began as a provider of Internet applications to real estate professionals. It became apparent that our customers valued the media exposure that the Internet offered them, but not all of the "technology" that we were offering. Many of our customers objected to our proposition that they purchase our templated web site in order to gain access to our networks. In addition, we were charging a fixed price to all customers regardless of the market they operate in or the size of their business.

We responded to our customers' needs and revamped our service offerings. We began to price our REALTOR.com[®] services based on the size of the market and the number of properties the customer displayed. For many of our customers this change led to substantial price increases over our former technology pricing. This change was reasonably well-accepted by our customer base.

In today's market, our real estate professional customers are facing a decline in their business and have to balance their marketing needs with their ability to pay. As a result, they are demanding products that perform and provide measurable results for their marketing spend. We are evaluating customer feedback and balancing that with the need for an improved consumer experience and are modifying our products and our pricing to be responsive to both.

The decline in consumer confidence and the resulting decline in consumer spending has caused many of our traditional consumer advertisers to reduce their spending. These economic conditions have caused the decline in our display and banner ad revenue. It could take considerable time before this product area yields meaningful growth, if at all. Achievement of significant growth will require that we introduce new targeted products that are both responsive to the advertisers' demands and are attractive to the consumer.

Acquisitions and Dispositions

On September 20, 2010, we acquired all of the outstanding shares of Threewide Corporation ("Threewide") for approximately \$13.1 million in cash. Threewide is the operator of ListHub, an online real estate listing syndicator and provider of advanced performance reporting solutions for brokers, real estate franchises, multiple listing services and real estate web sites. The acquisition cost has been allocated to the assets acquired, including intangible assets, and liabilities assumed based on their respective fair values.

In the second quarter of 2008, we decided to divest our Welcome Wagon® business. On June 22, 2009 we closed the sale of the business for a sales price of \$2.0 million. We received \$1.0 million in cash and a \$1.0 million promissory note. The principal balance of the note which was originally due on or before October 1, 2010 was paid in full in July 2010. The outstanding principal bore an interest rate of 7% per annum, with quarterly interest payments due commencing October 1, 2009. The transaction resulted in a gain on disposition of discontinued operations of \$1.2 million for the year ended December 31, 2009.

As part of the sale in 2002 of our ConsumerInfo division to Experian Holding, Inc. ("Experian"), \$10.0 million of the purchase price was put in escrow to secure our indemnification obligations (the "Indemnity Escrow"). Under the terms of the stock purchase agreement, our maximum potential liability for claims by Experian was capped at \$29.3 million less the balance in the Indemnity Escrow, which amount was approximately \$8.5 million on March 31, 2009. During 2008, Experian demanded \$29.3 million in indemnity payments. We denied liability and a bifurcated arbitration proceeding ensued to resolve the dispute. Subsequent to the completion of the first phase of the arbitration proceedings, on April 20, 2009, the parties settled the dispute and entered into a full release of all claims under which Experian received \$7.4 million from the Indemnity Escrow and we received the balance of the escrow of \$1.1 million, which is included in gain on disposition of discontinued operations for the nine months ended September 30, 2009.

Our Condensed Consolidated Financial Statements for all periods presented reflects the classification of our Welcome Wagon® business as discontinued operations. Accordingly, the revenue, operating expenses, and cash flows of this business has been excluded from the respective captions in the Condensed Consolidated Statements of Operations and Condensed

Consolidated Statements of Cash Flows and have been reported as "Income (loss) from discontinued operations," net of applicable income taxes of zero; and as "Net cash used in discontinued operations." Total revenue and income (loss) from discontinued operations are reflected below (in thousands):

		For the Three Months Ended September 30,		e Months Ended mber 30,
	2010	2009	2010	2009
Revenue	\$-	\$-	\$-	\$9,609
Total operating expenses	_	196	-	9,009
Impairment of long-lived assets	-	-	-	-
Restructuring charges	-	—	-	1,045
Loss from discontinued operations	\$-	\$(196)	\$-	\$(445)
Gain on disposition of discontinued operations	<u>\$-</u>	<u>\$-</u>	<u>\$-</u>	\$2,303

Critical Accounting Policies

Our discussion and analysis of our financial condition and results of operations is based upon our unaudited Condensed Consolidated Financial Statements, which have been prepared in accordance with U.S. generally accepted accounting principles. The preparation of these unaudited Condensed Consolidated Financial Statements requires us to make estimates and judgments that affect the reported amounts of assets, liabilities, revenues and expenses, and related disclosure of contingent assets and liabilities. On an ongoing basis, we evaluate our estimates, including those related to revenue recognition, uncollectible receivables, valuation of investments, intangible and other long-lived assets, stock-based compensation and contingencies. We base our estimates on historical experience and on various other assumptions that are believed to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. Actual results may differ from these estimates under different assumptions or conditions. There were no significant changes to our critical accounting policies during the nine months ended September 30, 2010, as compared to those policies disclosed in our Annual Report on Form 10-K for the fiscal year ended December 31, 2009.

Legal Contingencies

We are currently involved in certain legal proceedings, as discussed in Note 23, "Commitments and Contingencies–Legal Proceedings," to our Consolidated Financial Statements contained in Item 8 in our Annual Report on Form 10-K for the year ended December 31, 2009, and in Note 15, "Commitments and Contingencies" to our Unaudited Condensed Consolidated Financial Statements contained in Item 1 of Part I of this Form 10-Q. Because of the uncertainties related to both the amount and range of potential liability in connection with legal proceedings, we are unable to make a reasonable estimate of the liability that could result from unfavorable outcomes in our remaining pending litigation. As additional information becomes available, we will assess the potential liability related to our pending litigation and determine whether reasonable estimates of the liability can be made. Unfavorable outcomes, or significant estimates of our potential liability, could materially impact our results of operations and financial position.

Results of Operations

Three Months Ended September 30, 2010 and 2009

Revenue

Revenue decreased \$2.6 million, or 5%, to \$50.3 million for the three months ended September 30, 2010, compared to \$52.9 million for the three months ended September 30, 2009. The decrease in revenue was primarily due to a decrease in our New Homes and Top Producer® products. Our New Homes revenues were transferred to our unconsolidated joint venture in the fourth quarter of 2009 and therefore are not included in revenue for the three months ended September 30, 2010. Our Top Producer® 8i® subscriber base and associated revenues decreased over the prior year due to reduced spending by real estate professionals but was partially offset by improved revenues from the Market Snapshot® and Market BuilderTM products. We had slight declines in our REALTOR.com® products with lower Featured CommunityTM revenue directly related to reduced spending by our agent customers in response to the general economic conditions partially offset by increased revenues generated by our new Search Assist product which was launched in the latter part of 2009. We also experienced declines in the Rentals showcase listings revenues. Our revenue was also impacted by a decrease of \$0.4 million associated with providing product development service to the NAR in the three months ended September 30, 2009. We experienced an increase in our online display revenue due to an increased level of ad campaigns.

Cost of Revenue

Cost of revenue decreased \$1.2 million, or 10%, to \$10.8 million for the three months ended September 30, 2010,

Copyright © 2012 www.secdatabase.com. All Rights Reserved. Please Consider the Environment Before Printing This Document

compared to \$12.0 million for the three months ended September 30, 2009. The decrease was primarily due to decreased personnel related costs of \$0.6 million, lower depreciation expense of \$0.2 million, and other cost decreases of \$0.4 million.

Gross margin percentage increased to 79% for the three months ended September 30, 2010, compared to 77% for the three months ended September 30, 2009 primarily due to the decreased costs described above.

Operating Expenses

Sales and marketing. Sales and marketing expenses decreased \$0.2 million, or less than 1%, to \$18.6 million for the three months ended September 30, 2010, compared to \$18.8 million for the three months ended September 30, 2009. The decrease was primarily due to a decrease in online distribution costs of \$0.5 million, partially offset by an increase in personnel related costs of \$0.3 million.

Product and web site development. Product and web site development expenses increased \$1.2 million, or 16%, to \$8.9 million for the three months ended September 30, 2010, compared to \$7.7 million for the three months ended September 30, 2009. The increase was primarily due to an increase in consulting and personnel related costs as a result of incremental investments in our new technology platforms.

General and administrative. General and administrative expenses decreased \$5.3 million, or 33%, to \$10.9 million for the three months ended September 30, 2010, compared to \$16.2 million for the three months ended September 30, 2009. The decrease was primarily a result of a \$2.1 million decrease in non-cash stock based compensation due to the acceleration and modification of options upon the termination of one executive officer during the three months ended September 30, 2009, a decrease in legal fees of \$1.9 million and a decrease in personnel related costs, excluding non-cash stock based charges, of \$1.3 million.

Amortization of intangible assets. Amortization of intangible assets was \$0.1 million for the three months ended September 30, 2010 and 2009.

Restructuring Charges. In the third and fourth quarters of 2008, our Board of Directors approved restructuring and integration plans with the objective of eliminating duplicate resources and redundancies and implementing a new operating structure to lower total operating expenses. As a result of these plans, we incurred a restructuring charge from continuing operations of \$4.4 million for the year ended December 31, 2008. Included in this charge were lease obligations and related charges of \$3.0 million for the consolidation of our operations in Westlake Village, California and the vacancy of a portion of the leased facility.

During the third quarter of 2009, we entered into a new lease agreement for our Westlake Village facility. Under the terms of the lease, we were leasing only a portion of the facility but were continuing to occupy our current space in that facility until construction was completed on the new space. Our obligations under the old lease were terminated and, as a result, the remaining restructuring reserve was reversed, resulting in a \$1.2 million credit to restructuring charges for the three months ended September 30, 2009.

Stock-based compensation and charges. The following chart summarizes the stock-based compensation and charges that have been included in the following captions for each of the periods presented (in thousands):

		Three Months Ended September 30,	
	2010	2009	
Cost of revenue	\$44	\$54	
Sales and marketing	400	472	
Product and web site development	453	166	
General and administrative	962	3,030	
Total from continuing operations	\$1,859	\$3,722	

Stock-based compensation and charges increased \$1.9 million for the three months ended September 30, 2010, compared to the three months ended September 30, 2009, primarily due the acceleration and modification of options upon the termination of one executive officer in the third quarter of 2009.

Interest Income, Net

Interest income, net, decreased slightly for the three months ended September 30, 2010, compared to the three months ended September 30, 2009, primarily due to the sale of higher interest earning Auction Rate Securities ("ARS") in April 2010.

Other Income (expense), Net

Other expense, net for the three months ended September 30, 2010 consisted primarily of foreign exchange fluctuations. Other income, net of \$1.3 million for the three months ended September 30, 2009 was generated primarily from the sale of assets.

Income Taxes

We recorded an income tax benefit of \$0.4 million in the three months ended September 30, 2010 and income tax expense of \$0.1 million in the three months ended September 30, 2010, we recorded an income tax benefit of \$0.5 million as a result of a change in the valuation allowances resulting from the deferred tax liability established for the amortizable intangible assets acquired as part of a business combination partially offset by \$0.1 million of state income tax expenses and a deferred tax provision related to amortization of certain indefinite lived intangible assets. In the three months ended September 30, 2009, income tax expense included state income taxes and a deferred tax provision related to amortization of certain indefinite lived intangible assets.

Results of Operations

Nine Months Ended September 30, 2010 and 2009

Revenue

Revenue decreased \$13.8 million, or 8%, to \$148.6 million for the nine months ended September 30, 2010, compared to \$162.4 million for the nine months ended September 30, 2009. The decrease in revenue was primarily due to a decrease in our New Homes and REALTOR.com® products. Our New Homes revenues were transferred to our unconsolidated joint venture in the fourth quarter of 2009 and therefore are not included in revenue for the nine months ended September 30, 2010. We experienced lower Featured CommunityTM, listing enhancement and Featured HomesTM revenue on REALTOR.com directly related to reduced spending by our agent customers in response to the general economic conditions partially offset by increased revenues generated by our new Search Assist product which was launched in the latter part of 2009. Our Top Producer® 8i® subscriber base and associated revenues decreased over the prior year due to reduced spending by real estate professionals but was offset by improved revenues from the Market Snapshot® and Market BuilderTM products. We also experienced declines in the Rentals showcase listings revenues, and in lead generation revenues from movers on our Moving.com web site. Revenue was also impacted by a decrease of \$2.4 million associated with providing product development services to the NAR in the nine months ended September 30, 2009. We experienced an increase in our online display revenue due to an increased level of ad campaigns.

Cost of Revenue

Cost of revenue decreased \$4.7 million, or 13%, to \$32.8 million for the nine months ended September 30, 2010, compared to \$37.5 million for the nine months ended September 30, 2009. The decrease was primarily due to decreased costs of \$1.5 million associated with development services previously provided to the NAR during the nine months ended September 30, 2009, a decrease in personnel related costs of \$1.4 million, a decrease in product fulfillment costs of \$1.1 million associated with our featured products, decreased depreciation expense of \$0.6 million, and other cost decreases of \$0.1 million.

Gross margin percentage increased to 78% for the nine months ended September 30, 2010, compared to 77% for the nine months ended September 30, 2009 primarily due to the decreased costs described above.

Operating Expenses

Sales and marketing. Sales and marketing expenses decreased \$5.1 million, or 8%, to \$55.8 million for the nine months ended September 30, 2010, compared to \$60.9 million for the nine months ended September 30, 2009. The decrease was primarily due to a decrease in online distribution costs of \$3.7 million, a decrease in personnel related costs of \$0.9 million due to lower sales commissions, and other cost decreases of \$0.5 million.

Product and web site development. Product and web site development expenses increased \$5.0 million, or 24%, to \$25.5 million for the nine months ended September 30, 2010, compared to \$20.5 million for the nine months ended September 30, 2009. The increase was primarily due to an increase in consulting and personnel related costs as a result of incremental investments in our new technology platforms.

General and administrative. General and administrative expenses decreased \$18.8 million, or 37%, to \$32.4 million for the nine months ended September 30, 2010, compared to \$51.2 million for the nine months ended September 30, 2009. The

decrease was primarily a result of a \$10.5 million decrease in non-cash stock based compensation primarily due to the acceleration and modification of options upon the termination of three executive officers and restricted stock awards and options granted to our Chief Executive Officer that were immediately vested during the nine months ended September 30, 2009. Additionally, there was a \$4.0 million decrease in personnel related costs, excluding non-cash stock based compensation, a \$3.8 million decrease in legal fees, a \$1.1 million decrease in bad debt expense and a \$0.6 million decrease in facilities costs due to the new lease in Westlake Village, California. These cost decreases were partially offset by an increase in consulting costs of \$1.1 million and other cost increases of \$0.1 million.

Amortization of intangible assets. Amortization of intangible assets was \$0.3 million and \$0.4 million for the nine months ended September 30, 2010 and 2009, respectively.

Restructuring Charges. In the third and fourth quarters of 2008, our Board of Directors approved restructuring and integration plans with the objective of eliminating duplicate resources and redundancies and implementing a new operating structure to lower total operating expenses. As a result of these plans, we incurred a restructuring charge from continuing operations of \$4.4 million for the year ended December 31, 2008. Included in this charge were lease obligations and related charges of \$3.0 million for the consolidation of our operations in Westlake Village, California and the vacancy of a portion of the leased facility.

During the third quarter of 2009, we entered into a new lease agreement for our Westlake Village facility. Under the terms of the lease, we were leasing only a portion of the facility but were continuing to occupy our current space in that facility until construction was completed on the new space. Our obligations under the old lease were terminated and, as a result, the remaining restructuring reserve was reversed, resulting in a \$1.2 million credit to restructuring charges for the nine months ended September 30, 2009.

Stock-based compensation and charges. The following chart summarizes the stock-based compensation and charges that have been included in the following captions for each of the periods presented (in thousands):

		Nine Months Ended September 30,	
	2010	2009	
Cost of revenue	\$136	\$137	
Sales and marketing	1,240	1,349	
Product and web site development	1,243	493	
General and administrative	3,146	13,668	
Total from continuing operations	\$5,765	\$15,647	

Stock-based compensation and charges decreased \$9.9 million for the nine months ended September 30, 2010 compared to the nine months ended September 30, 2009 primarily due to the acceleration and modification of options upon termination for three executive officers and restricted stock awards and options granted to our Chief Executive Officer that were immediately vested during the nine months ended September 30, 2009.

Interest Income, Net

Interest income, net, remained relatively constant for the nine months ended September 30, 2010 and 2009, respectively.

Impairment of Auction Rate Securities

In April 2010, we completed a sale of our entire portfolio of ARS for \$109.8 million (par value of \$129.4 million) to a broker in a secondary market. As a result of the sale, an other-than-temporary loss of \$19.6 million was recorded for the nine months ended September 30, 2010. See further discussion under *Liquidity and Capital Resources* below.

Other Income (expense), Net

Other expense, net of \$1.1 million for the nine months ended September 30, 2010, consisted primarily of the transaction fees associated with the sale of our portfolio of ARS. Other income, net of \$1.8 million for the nine months ended September 30, 2009 primarily resulted from a \$1.3 million gain on the sale of assets and \$0.5 million in other income related to the revaluation of an embedded derivative liability resulting from the issuance of convertible preferred stock in December 2005.

Income Taxes

We recorded an income tax benefit of \$0.3 million in the nine months ended September 30, 2010 and income tax expense of \$0.2 million in the nine months ended September 30, 2009. In the nine months ended September 30, 2010, we recorded an income tax benefit of \$0.5 million as a result of a change in the valuation allowances resulting from the deferred tax liability

established for the amortizable intangible assets acquired as part of a business combination partially offset by \$0.2 million of state income tax expenses and a deferred tax provision related to amortization of certain indefinite lived intangible assets. In the nine months ended September 30, 2009, income tax expense included state income taxes and a deferred tax provision related to amortization of certain indefinite lived intangible assets.

Liquidity and Capital Resources

Net cash provided by continuing operating activities of \$17.7 million for the nine months ended September 30, 2010 was attributable to the net loss of \$17.2 million, plus non-cash expenses including depreciation, amortization of intangible assets, provision for doubtful accounts, impairment of auction rate securities, stock-based compensation and charges, earnings of unconsolidated joint venture, other non-cash items and changes in operating assets and liabilities, aggregating to \$34.9 million.

Net cash provided by continuing operating activities of \$10.2 million for the nine months ended September 30, 2009 was attributable to the net loss from continuing operations of \$5.6 million, plus non-cash expenses including depreciation, amortization of intangible assets, provision for doubtful accounts, gain on sale of assets, stock-based compensation and charges, change in market value of embedded derivative liability and other non-cash items, aggregating to \$22.9 million, partially offset by changes in operating assets and liabilities of \$7.1 million.

Net cash provided by continuing investing activities of \$90.6 million for the nine months ended September 30, 2010 was primarily attributable to proceeds from the sale of auction rate securities of \$109.8 million and payments received on note for sale of business of \$1.0 million, partially offset by acquisitions, net of cash acquired of \$12.4 million and capital expenditures of \$7.8 million.

Net cash used in continuing investing activities of \$6.4 million for the nine months ended September 30, 2009 was due to capital expenditures of \$7.7 million, partially offset by proceeds from sales of fixed assets of \$1.3 million.

Net cash used in financing activities of \$59.5 million for the nine months ended September 30, 2010 was primarily attributable to principal payments on our line of credit of \$129.3 million offset by proceeds from our line of credit of \$64.7 million and other miscellaneous cash used of \$0.2 million. There were cash proceeds from the exercise of stock options of \$4.5 million, reductions in restricted cash balances of \$0.5 million and proceeds from loans payable of \$0.3 million.

Net cash provided by financing activities of \$3.2 million for the nine months ended September 30, 2009 was attributable to reductions is restricted cash of \$2.7 million and proceeds from the exercise of stock options of \$1.9 million, partially offset by tax withholdings related to net share settlements of restricted stock awards of \$1.1 million and payments on capital lease obligations of \$0.3 million.

We have generated positive operating cash flows in each of the last two years. Our material financial commitments consist of those under operating lease agreements, our operating agreement with the NAR and various web services and content agreements. Additionally, under our Series B Preferred Stock Agreement, beginning in November 2010, we are obligated to pay quarterly cash dividends of 3.5% per annum of the original price per share or approximately \$4.1 million per annum. We believe that our existing cash and any cash generated from operations will be sufficient to fund our working capital requirements, capital expenditures and other obligations for the foreseeable future.

Prior to April 2010, we had investments consisting of high-grade (primarily AAA rated) student loan auction rate securities issued by student loan funding organizations, which loans were 97% guaranteed under FFELP (Federal Family Education Loan Program). These ARS were intended to provide liquidity via an auction process that resets the interest rate, generally every 28 days, allowing investors to either roll over their holdings or sell them at par. In February 2008, auctions for our investments in these securities failed to settle on their respective settlement dates. Consequently, the investments were not liquid and we were not going to be able to access these funds until a future auction of these investments was successful, the securities matured or a buyer was found outside of the auction process. Maturity dates for these ARS investments range from years 2030 to 2047 with principal distributions occurring on certain securities prior to maturity.

As of December 31, 2009, we had recorded a temporary loss related to the ARS of \$17.6 million that was included in Other Comprehensive Income on our Consolidated Balance Sheet. At a board meeting on March 24, 2010, the Board of Directors and Management discussed the recent passage of the Health Care Reform Bill that contained a provision eliminating FFELP, a significant change in student loan funding. In management's opinion, this change, along with other market factors, created additional uncertainty in the student loan auction rate securities market. As a result, the Board of Directors and Management changed its intent, which had been to hold these securities, and decided to sell the entire portfolio of ARS and, thereafter, we began to actively market the sale to third parties. We review our potential investment impairments in accordance with ASC 320 "Investment – Debt and Equity Securities" and the related guidance issued by the FASB and SEC in order to determine the classification of the impairment as "temporary" or "other-than-temporary." A temporary impairment charge results in an unrealized loss being recorded in the other comprehensive income (loss) component of stockholder's equity. An otherthan-temporary impairment charge is recorded as a realized loss in the Condensed Consolidated Statement of Operations and reduces net income (loss) for the applicable accounting period. The differentiating factors between temporary and other-than-temporary impairment are primarily the length of the time and the extent to which the market value has been less than cost, the financial condition and near-term prospects of the issuer, and the ability and intent of the holder to hold the investment until maturity or its value recovers. Prior to March 24, 2010, we had not intended to sell nor was it not more likely than not that we would be required to sell before the recovery of our amortized cost basis and, as such, the loss was considered temporary. On March 24, 2010, as indicated above, we changed our intent to hold the ARS and, therefore, the impairment was reclassified to an other-than-temporary loss.

In April 2010, we completed a sale of the entire portfolio of ARS for \$109.8 million (par value of \$129.4 million) to a broker in a secondary market. As a result of the sale, an other-than-temporary loss of \$19.6 million was recorded as Impairment of Auction Rate Securities in our Condensed Consolidated Statement of Operations for the nine months ended September 30, 2010. The transaction costs of approximately \$1.0 million associated with this transaction has been recorded as other expense for the nine months ended September 30, 2010.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

Interest Rate Risk

Market risk represents the risk of loss that may impact our financial position, results of operations or cash flows due to adverse changes in financial and commodity market prices and rates. We are exposed to market risk primarily in the area of changes in United States interest rates and conditions in the credit markets. In April 2010, we paid down the line of credit balance of \$64.2 million and it expired in May 2010. We do not have any material foreign currency or other derivative financial instruments. Under our current policies, we do not use interest rate derivative instruments to manage exposure to interest rate changes. We attempt to increase the safety and preservation of our invested principal funds by limiting default risk, market risk and reinvestment risk. We mitigate default risk by investing in investment grade securities.

Item 4. Controls and Procedures

As of the end of the period covered by this report, we carried out an evaluation, under the supervision and with the participation of our Chief Executive Officer and Chief Financial Officer, of the effectiveness of our disclosure controls and procedures pursuant to Rule 13a-15 of the Securities Exchange Act of 1934 (the "Exchange Act"). Based upon that evaluation, the Chief Executive Officer and Chief Financial Officer concluded that our disclosure controls and procedures are effective to ensure that information required to be disclosed by us in reports that we file or submit under the Exchange Act (i) is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms and (ii) is accumulated and communicated to our management including our Chief Executive Officer and Chief Financial Officer, as appropriate, to allow timely decisions regarding required disclosure.

There were no changes in our internal control over financial reporting during the period covered by this report that materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II. OTHER INFORMATION

Item 1. Legal Proceedings

We are currently involved in certain legal proceedings, as discussed in Note 23, "Commitments and Contingencies- Legal Proceedings", to our Consolidated Financial Statements contained in Item 8 in our Annual Report on Form 10-K for the year ended December 31, 2009 ("Annual Report") and in Note 15, "Commitments and Contingencies," to the Unaudited Condensed Consolidated Financial Statements contained in Item 1 of Part I of this Form 10-Q. As of the date of this Form 10-Q and except as disclosed in Note 23 to the Consolidated Financial Statements in our Annual Report and in Note 15 to the Unaudited Condensed Consolidated Financial Statements in this Form 10-Q, the Company is not a party to any other litigation or administrative proceedings that management believes will have a material adverse effect on the Company's business, results of operations, financial condition or cash flows, and there have been no material developments in the litigation or administrative proceedings described in those notes.

Item 1A. Risk Factors

You should consider carefully the risk factors presented in our Annual Report on Form 10-K for the year ended December 31, 2009, and other information included or incorporated by reference in this Form 10-Q. The risks and uncertainties described below are not the only ones we face. Additional risks and uncertainties not presently known to us or that we deem to be currently immaterial also may impair our business operations. If any of the stated risks actually occur, our business, financial condition and operating results could be materially adversely affected.

Risks Related to our Business

NAR relationship

We have completed our discussions and the mediation process concerning questions relating to the 1996 Operating Agreement (as referenced in our reports on Form 10-Q for, respectively, the first and second quarters of 2010, in each case at *Item 1A – Risk Factors*), have agreed with the NAR on an amendment to the 1996 Operating Agreement (as reported in our current report on Form 8-K filed September 16, 2010), and continue to value our relationship with and opportunities for productive dialogue with the NAR.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

None.

Item 3. Defaults Upon Senior Securities

None.

Item 4. [Removed and Reserved]

None.

Item 5. Other Information

None.

Item 6. Exhibits

Exhibits

- 10.73 Amendment dated September 10, 2010, to the Operating Agreement dated November 26, 1996, between RealSelect, Inc., a wholly owned subsidiary of the Company, and Realtors Information Network, Inc., a wholly-owned subsidiary of the National Association of REALTORS. (Incorporated by reference to Exhibit 10.1 to our Current Report on Form 8-K filed September 16, 2010.)
- 10.74 Agreement and Plan of Merger entered into by Move Sales, Inc. (a wholly owned subsidiary of the Company) and MSI TC Merger Sub (a wholly owned subsidiary of Move Sales, Inc.) on September 17, 2010, with Threewide Corporation and, as the Threewide Corporation equity-holders' representatives, both J. Rudy Henley in his capacity as manager of West Virginia Capital Management LLC and William Rice.
- 31.1 Certification of Chief Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 31.2 Certification of Chief Financial Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 32.1 Certification of Chief Executive Officer pursuant to 18 U.S.C. Section 1350 as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
- 32.2 Certification of Chief Financial Officer pursuant to 18 U.S.C. Section 1350 as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

MOVE, INC.

By: /s/ STEVEN H. BERKOWITZ

Steven H. Berkowitz Chief Executive Officer

By: /s/ ROBERT J. KROLIK

Robert J. Krolik Chief Financial Officer

Date: November 5, 2010

EXHIBIT INDEX

Exhibit	
Number	Description
10.73	Amendment dated September 10, 2010, to the Operating Agreement dated November 26, 1996, between RealSelect, Inc., a wholly
	owned subsidiary of the Company, and Realtors Information Network, Inc., a wholly-owned subsidiary of the National Association of REALTORS. (Incorporated by reference to Exhibit 10.1 to our Current Report on Form 8-K filed September 16, 2010.)
10.74	Agreement and Plan of Merger entered into by Move Sales, Inc. (a wholly owned subsidiary of the Company) and MSI TC Merger

- Sub (a wholly owned subsidiary of Move Sales, Inc.) on September 17, 2010, with Threewide Corporation and, as the Threewide Corporation equity-holders' representatives, both J. Rudy Henley in his capacity as manager of West Virginia Capital Management LLC and William Rice.
- 31.1 Certification of Chief Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 31.2 Certification of Chief Financial Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 32.1 Certification of Chief Executive Officer pursuant to 18 U.S.C. Section 1350 as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
- 32.2 Certification of Chief Financial Officer pursuant to 18 U.S.C. Section 1350 as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

AGREEMENT AND PLAN OF MERGER

This Agreement and Plan of Merger (the "Agreement") is entered into as of September 15, 2010 by and among Move Sales, Inc., a Delaware corporation ("Purchaser"); Threewide Corporation, a Delaware corporation ("Threewide"); MSI TC Merger Sub, Inc., a Delaware corporation and a wholly-owned subsidiary of Purchaser ("Merger Sub"); and William Rice, an individual resident of West Virginia, and J. Rudy Henley in his capacity as a manager of West Virginia Capital Management LLC, a West Virginia limited liability company (collectively, the "Equityholders' Representative"). Purchaser, Threewide, Merger Sub and the Equityholders' Representative are each referred to herein as a "Party" and collectively as the "Parties".

RECITALS:

WHEREAS, the board of directors of each of the Purchaser, Threewide and Merger Sub has approved, and deems it fair, advisable and in the best interests of their respective companies and stockholders to consummate the merger of Merger Sub with and into Threewide upon the terms and subject to the conditions set forth herein (collectively, the "Merger"); and

WHEREAS, in furtherance of the Merger, the board of directors of each of the Purchaser, Threewide and Merger Sub has approved this Agreement and the transactions contemplated herein upon the terms and subject to the conditions set forth in this Agreement; and

WHEREAS, immediately following the execution of this Agreement, the board of directors of Threewide shall deliver to the Requisite Stockholders this Agreement and the Stockholder Consent, which approves among other things the adoption of this Agreement and the transactions contemplated hereby to be performed by Threewide; and

WHEREAS, it is the intention of the Parties that, upon consummation of the Merger pursuant to this Agreement, Threewide shall be a wholly-owned subsidiary of the Purchaser.

AGREEMENT:

In consideration of the Recitals and of the agreements and covenants herein contained, and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties hereby agree as follows:

1. DEFINITIONS

As used herein, the following terms shall have the following meanings unless the context otherwise requires:

"Affiliate" means with respect to a specified Person, a Person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, the Person specified or one or more Affiliates thereof. For the purposes of this definition, "<u>controlled</u>" and "<u>control</u>" mean the possession, directly or indirectly, of the power to direct the management and policies of a Person whether through the ownership of voting securities, contract or otherwise.

"Agreement" has the meaning set forth in the introductory paragraph, above.

"Appraisal Rights" means the rights of stockholders of a corporation in a merger to demand the payment of a fair price for their shares, as determined independently, as set forth in Section 262 of the DGCL.

"Base Consideration" has the meaning provided in Section 2.3(a).

"Basket" has the meaning provided in Section 6.2(b).

"Business" means the business operated by Threewide as of the date of this Agreement.

"Cap" has the meaning provided in Section 6.2(c).

"Certificate of Merger" means the certificate of merger in the form attached as Exhibit A.

"Closing" has the meaning set forth in Section 2.1.

"Closing Date" has the meaning set forth in Section 2.1.

"Code" has the meaning set forth in Section 3.9(b).

"Company Certificate" has the meaning set forth in Section 2.2(e).

"Continuing Employees" has the meaning provided in Section 5.2.

"DGCL" means the Delaware General Corporation Law.

"Disclosure Schedules" has the meaning set forth in Section 3.

"Dissenting Shares" has the meaning provided in Section 2.7.

"Effective Time" has the meaning provided in Section 2.2(b).

"Employee Pension Benefit Plan" has the meaning set forth in ERISA Section 3(2).

"Employee Welfare Benefit Plan" has the meaning set forth in ERISA Section 3(1).

"Employment Agreements" means the Employment Agreement between Threewide and Luke Glass in the form attached hereto as Exhibit G-1 executed on the date hereof to be effective upon the Closing; the Employment Agreement between Threewide and Mark T. Wise in the form attached hereto as Exhibit G-2 executed on the date hereof to be effective upon the Closing; the Employment Agreement between Threewide and Rob Reid in the form attached hereto as Exhibit G-3 executed on the date hereof to be effective upon the Closing; and the Employment Agreement between Threewide and Celeste Starchild in the form attached hereto as Exhibit G-4 executed on the date hereof to be effective upon the Closing.

"Equityholder" means a Stockholder or a Warrantholder. "Equityholders" means, collectively, the Stockholders and Warrantholders.

- 2 -

"Equityholders' Representative" has the meaning provided in the introductory paragraph.

"ERISA" has the meaning set forth in Section 3.16(b).

"Escrow Agreement" shall have the meaning set forth in Section 2.4(a)(i).

"Escrow Agent" shall mean U.S. Bank N.A.

"Escrow Amount" has the meaning set forth in Section 2.4(a)(i).

"Expenses Cap" has the meaning set forth in Section 2.3(a)(iii).

"Expenses Holdback" has the meaning provided in Section 2.4(a)(ii).

"Expenses Payment" has the meaning provided in Section 2.4(c)(ii).

"GAAP" means United States generally accepted accounting principles, as in effect from time to time.

"Governmental Entity" means: (i) any national government, political subdivision thereof or local jurisdiction therein; (ii) any instrumentality, board, commission, court or agency of any of the foregoing, however constituted; and (iii) any association, organization or institution of which any of the foregoing is a member or to whose jurisdiction any thereof is subject or in whose activities any of the above is a participant.

"Indemnified Party" has the meaning set forth in Section 6.3(a).

"Indemnifying Party" has the meaning set forth in Section 6.3(a).

"Intellectual Property" means all formulas, trade secrets, trademarks, trade names, domain names, codes, patents, copyrights, inventions and discoveries that are being used by Threewide, are owned or licensed by Threewide, or are registered or filed in the name of Threewide (whether exclusively, jointly with another Person, or otherwise), including any and all recipes, formulas or inventions in process (whether or not reduced to practice) or any patent or trademark applications filed by Threewide.

"Letter of Transmittal" has the meaning provided in Section 2.5(a).

"Management Carveout" has the meaning provided in Section 2.3(a).

"Material Adverse Change" or "Material Adverse Effect" means any event, circumstance, development, change or effect that has had or could have a material adverse effect on the business, condition (financial or otherwise), properties, assets, liabilities, prospects or results of operations of Threewide.

"Merger" has the meaning provided in the first recital.

"Merger Consideration" has the meaning provided in Section 2.3(a).

- 3 -

"Merger Sub" has the meaning provided in the introductory paragraph.

"Option Payment" has the meaning provided in Section 2.6(a).

"Option Termination Agreements" means the Option Termination Agreements between Threewide and the holders of the options to acquire Stock in the form attached hereto as Exhibit H.

"Parties" or "Party" shall mean Merger Sub, Threewide, and Purchaser or any of them.

"Person" means any individual, corporation (including any non-profit corporation), partnership, limited liability company, joint venture, estate, trust, association, organization, labor union or other entity or Governmental Entity.

"Pro Rata Share" means the amounts and percentages set forth on Exhibit B.

"Purchaser" has the meaning set forth in the introductory paragraph.

"Purchaser Indemnified Parties" has the meaning provided in Section 6.2(a).

"Requisite Stockholders" shall mean the holders of at least: (a) a majority of Threewide's issued and outstanding common stock (including the preferred stock on an as-converted basis), (b) a majority of Threewide's issued and outstanding Series A Preferred Stock, (c) a majority of Threewide's issued and outstanding Series A-1 Preferred Stock, (d) two-thirds of Threewide's issued and outstanding Series B Preferred Stock, and (e) two-thirds of Threewide's issued and outstanding Series C Preferred Stock (including such stock as may be purchased by the exercise of any Warrant).

"Stock" means the capital stock, of any class, of Threewide.

"Stockholder" means the holder of any Stock. "Stockholders" means, collectively, the holders of any Stock.

"Stockholder Consent" means the stockholder consent in the form attached hereto as Exhibit I.

"Stock Option Plan" means the Threewide 2001 Stock Option Plan, as amended, attached hereto as Exhibit D.

"Surviving Corporation" has the meaning provided in Section 2.2(a).

"Tax" or "Taxes" means any federal, state, local, or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental (including taxes under Code §59A), customs duties, capital stock, franchise, profits, withholding, social security (or similar), unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, alternative or add-on minimum, estimated, or other tax of any kind whatsoever, when computed on a separate or consolidated,

- 4 -

unitary or combined basis or in any other manner, including any interest, penalty, or addition thereto, whether disputed or not.

"Third-Party Claim" has the meaning set forth in Section 6.3(c).

"Threewide" has the meaning set forth in the introductory paragraph.

"Threewide Financials" has the meaning set forth in Section 3.5.

"To Threewide's knowledge" and similar phrases shall mean the actual knowledge of Luke Glass, Mark Wise, Rob Reid, Celeste Starchild, the Equityholders' Representative and each member of Threewide's board of directors (where the actual knowledge of each of such persons is imputed to all other of such persons), who, for the avoidance of doubt, are listed on Schedule 3.2(c).

"Transaction Expenses" means fees, costs, and expenses incurred pursuant to the transactions contemplated by this Agreement, including, but not limited to, the fees and expenses of legal counsel, accountants, investment bankers, consultants and other experts.

"Warrants" means the warrants to acquire any Stock.

"Warrantholder" means the holder of any Warrants. "Warrantholders" means, collectively, the holders of any Warrants.

2. MERGER

2.1 The Closing. The closing of the transactions contemplated by this Agreement (the "Closing") shall take place at Threewide's offices at 2 p.m. on September 20, 2010 or such other date as the Parties may mutually agree on which all conditions precedent to the Closing described in Section 7 are met or waived (the date of the Closing is referred to herein as the "Closing Date").

2.2 The Merger.

(a) At the Effective Time, and subject to and upon the terms and conditions of this Agreement, the Certificate of Merger and the applicable provisions of the DGCL, Merger Sub shall be merged with and into Threewide and the separate corporate existence of Merger Sub shall cease, and Threewide shall continue as the surviving corporation under the laws of the State of Delaware (the "Surviving Corporation").

(b) The Merger shall become effective when: (i) the Closing has occurred; and (ii) the Certificate of Merger, executed in accordance with the DGCL, is duly filed with the Secretary of State of Delaware, or such time as the Parties may agree and specify in the Certificate of Merger (such time of effectiveness, the "Effective Time").

(c) From and after the Effective Time, the Merger shall have the effects provided for in Section 259 of the DGCL.

(d) From and after the Effective Time: (i) the certificate of incorporation of Merger Sub, as in effect immediately prior to the Effective Time, shall be the certificate of incorporation of the Surviving Corporation (except that the name of the Surviving Corporation shall be Threewide Corporation and the provision in the certificate of incorporation of Merger Sub naming its incorporator shall be omitted) until thereafter amended as provided by law and such certificate of incorporation; (ii) the bylaws of Merger Sub, as in effect immediately prior to the Effective Time, shall be the bylaws of the Surviving Corporation until thereafter amended as provided by law, by the certificate of incorporation until thereafter amended as provided by law, by the certificate of incorporation of the Surviving Corporation or by such bylaws; and (iii) the directors and officers of Merger Sub immediately prior to the Effective Time shall be the directors and officers, respectively, of the Surviving Corporation, each of such directors and officers to hold office, subject to the applicable provisions of the certificate of incorporation and bylaws of the Surviving Corporation, until their respective successors shall be duly elected or appointed and qualified.

(e) At the Effective Time, by virtue of the Merger and without any further action by the Parties, each share of Stock issued and outstanding immediately prior to the Effective Time and each Warrant issued and outstanding immediately prior to the Effective Time shall be converted into the right to receive a portion of the Merger Consideration as set forth in this Agreement upon surrender of the certificate or agreement formerly representing such shares of Stock or Warrants (each a "Company Certificate"), payable at the times and in the manner, and subject to the terms and conditions, provided in this Agreement. All such converted Stock and Warrants, when so converted, shall no longer be outstanding and shall be deemed to be cancelled and retired and shall cease to exist, and each holder of any Stock or Warrants shall cease to have any rights with respect to any Company Certificate in accordance with this Agreement.

2.3 Consideration.

(a) The total consideration to be paid in connection with the transactions contemplated by this Agreement shall be equal to \$13,000,000 (the "Base Consideration"), as increased or decreased on a dollar-for-dollar basis for the cumulative net adjustments required by the following (as adjusted, the "Merger Consideration"):

i. The Merger Consideration shall be decreased by the Option Payment, which amount shall be payable to certain members of Threewide's management team pursuant to Section 2.9;

ii. The Merger Consideration shall be decreased by \$994,863.31 (the "Management Carveout"), which amount shall also be payable to certain members of Threewide's management team pursuant to Section 2.9; and

iii. The Merger Consideration shall be increased by the amount of Threewide's Transaction Expenses that have not been paid by Threewide prior to Closing, which are to be set forth on Schedule 2.9(e) to be delivered at Closing, which such amount shall not exceed \$100,000 less the amount of Threewide's

- 6 -

Transaction Expenses paid by it prior to Closing, which amount shall also be set forth on Schedule 2.9(e) (the "Expenses Cap").

(b) On the Closing Date, each share of Stock and each Warrant shall be converted such that, at the Effective Time, each Equityholder shall be entitled to receive such Equityholder's Pro Rata Share of the Merger Consideration (less the amount of the Holdbacks) by bank check or wire transfer in accordance with the terms and conditions of this Agreement. The Holdbacks will be applied as set forth in Section 2.4(a). Each Equityholder's Pro Rata Share of the Merger Consideration and the Holdbacks shall be calculated in accordance with the methodology set forth on Exhibit B, provided, however, that if there are any Dissenting Shares, the percentages on Exhibit B shall be recalculated as if such Dissenting Shares did not exist. At the Closing, Threewide shall deliver to Purchaser Exhibit B updated as of the Closing, and, in paying the Equityholders' Pro Rata Shares of the Merger Consideration, Purchaser shall be entitled to rely upon the calculations and amounts set forth on such updated Exhibit B without any investigation or independent review.

(c) Each issued and outstanding share of common stock of Merger Sub shall be converted into one share of common stock of the Surviving Corporation.

2.4 Escrow and Holdbacks.

(a) Purchaser shall withhold payment of a portion of the Merger Consideration and apply it as follows (collectively, the "Holdbacks"):

i. On the Closing Date, Purchaser shall deposit an amount equal to \$3,000,000 (the "Escrow Amount") with the Escrow Agent in the form of a wire transfer of immediately available funds into an account designated prior to the Closing, to be held and disbursed pursuant to the terms and provisions of the Escrow Agreement attached hereto as Exhibit C (the "Escrow Agreement"); and

ii. An amount equal to \$100,000 for payment or reimbursement of the Equityholders' Representative's expenses incurred in connection with his duties under this Agreement (the "Expenses Holdback"). The Expenses Holdback will be paid to the Equityholders' Representative in the form of a wire transfer of immediately available funds into an account designated prior to the Closing.

(b) Each Stockholder (except to the extent such Stockholder's shares are Dissenting Shares) and Warrantholder shall be deemed to have contributed to the Holdbacks pro rata in an amount equal such Equityholder's Pro Rata Share thereof. Stockholders holding Dissenting Shares shall not be deemed to have contributed to the Holdbacks to the extent related to such Dissenting Shares.

(c) Each Equityholder's Pro Rata Share shall be available from time to time, as follows:

i. In respect of the Escrow Amount, to satisfy any indemnification obligations pursuant to Section 6, including without limitation

- 7 -

any underlying claims in respect of any Appraisal Rights and any other costs directly attributable to such claims (an "Indemnity Payment"). Except as specifically set forth in Section 6, any Indemnity Payments shall be satisfied solely by reducing each Equityholder's Pro Rata Share in respect of the Escrow Amount by an amount equal to the Equityholder's Pro Rata Share of such Indemnity Payment, to the extent such resulting amount does not exceed such Equityholder's Pro Rata Share in respect of the Escrow Amount.

ii. In respect of the Expenses Holdback, to pay for or reimburse the Equityholders' Representative for his expenses incurred in connection with his duties under this Agreement (each, an "Expenses Payment"). Any Expenses Payments shall be satisfied solely by reducing each Equityholder's Pro Rata Share in respect of the Expenses Holdback by an amount equal to such Equityholder's Pro Rata Share in respect of the Expenses Holdback by an amount equal to such Equityholder's Pro Rata Share in respect of the Expenses Holdback. Once the Equityholders' Representative has determined that no more Expenses Payments will be paid, then the Equityholders' Representative will distribute all Equityholders' Pro Rata Shares in respect of the remaining Expenses Holdback.

(d) The balance, if any, of the Equityholders' Pro Rata Shares in respect of the Escrow Amount shall be released and paid over to the Equityholders' Representative for distribution to the Equityholders in accordance with the Escrow Agreement (for the avoidance of doubt, only Warrantholders and Stockholders who have irrevocably forgone or lost their Appraisal Rights will be entitled to share in such distribution).

2.5 Exchange of Certificates.

(a) Immediately following the execution of this Agreement and in no event more than three days after the date hereof, the Equityholders' Representative shall obtain from (i) the Requisite Stockholders, (A) the Stockholder Consent in accordance with the applicable provisions of the DGCL (including Sections 228 and 251 of the DGCL) and executed by the Requisite Stockholders, (B) an executed letter of transmittal in the form attached hereto as Exhibit J (a "Letter of Transmittal"), and (C) the Company Certificates representing all of the shares of Stock held by the Requisite Stockholders, and (iii) the Warrantholders, (A) an executed Letter of Transmittal and (B) the Company Certificates representing all of the Warrants held by the Warrantholders. As soon as reasonably practicable following the Closing Date, Threewide shall mail, fax, send electronically or send by overnight courier to each holder of record of one or more shares of Stock (not otherwise obtained pursuant to the previous sentence) a Letter of Transmittal.

(b) Following the Closing (or at the Closing, in the case of the Requisite Stockholders and the Warrantholders) and surrender of a Company Certificate representing the shares of Stock held by a Stockholder and/or the Warrants held by a Warrantholder, together with a Letter of Transmittal, duly executed, the holder of the Stock and/or Warrants represented by such Company Certificate shall be entitled to receive in exchange therefor the portion of the Merger Consideration into which such Stock and/or Warrants shall have been converted pursuant



to this Section 2, and the Company Certificate so surrendered to the Surviving Corporation shall be cancelled. If payment of such portion of the Merger Consideration is to be made to a Person other than the Person in whose name such Company Certificate is registered, it shall be a condition of payment that the Company Certificate so surrendered shall be properly endorsed or shall be otherwise in proper form for transfer and that the Person requesting such payment shall have paid any transfer and other Taxes required by reason of the payment of the portion of the Merger Consideration to a Person other than the registered holder of the Stock and/or Warrant or shall have established to the satisfaction of the Surviving Corporation that such Tax either has been paid or is not applicable. In the event that any Company Certificate shall have been lost, stolen or destroyed, the Purchaser shall pay in exchange therefor, upon making of an affidavit of that fact by the holder thereof in form and substance acceptable to the Purchaser, the portion of the Merger Consideration and as a condition precedent to the issuance thereof, require the delivery of a suitable indemnity agreement by the owner of such lost, stolen or destroyed Company Certificate, and may at its option require the indemnity to be secured by a bond in form and substance satisfactory to Purchaser.

(c) The Purchaser may pay over to the Surviving Corporation any portion of the Merger Consideration remaining 180 days after the Effective Time, and thereafter all former Equityholders shall be entitled to look to the Surviving Corporation (subject to abandoned property, escheat, and other similar laws) as general creditors thereof with respect to the cash payable upon surrender of their Company Certificates. None of the Purchaser, Merger Sub or the Surviving Corporation shall be liable to any Equityholder for any portion of the Merger Consideration delivered to a public official pursuant to any applicable abandoned property, escheat or similar law. On the date hereof, the stock transfer books of Threewide shall be closed to transfers of Stock and exercise of Warrants outstanding immediately prior to the date hereof, and thereafter there shall be no further registration of transfers of such Stock or exercise of Warrants on the records of Threewide, unless this Agreement is terminated.

2.6 Stock Options and Warrants.

(a) All outstanding options to acquire Stock pursuant to the Stock Option Plan outstanding as of the date hereof are set forth on Exhibit D (along with a copy of the Stock Option Plan). At the Closing, such options shall be terminated and Threewide shall pay the amounts therefor set forth on Exhibit D (the "Option Payment") in consideration for the termination of such options in accordance with and pursuant to the Option Termination Agreements.

(b) All Warrants outstanding as of the date hereof are set forth on Exhibit E. At the Closing, the Warrants shall be converted into the right to receive a portion of the Merger Consideration (less the amount of the Holdbacks) as set forth on Exhibit B. The Holdbacks will be applied as set forth in Section 2.4(a).

2.7 Appraisal Rights and Dissenting Shares. Immediately following the Closing but in any event within 10 days of the Closing Date, the Surviving Corporation shall mail, fax, send electronically or send by overnight courier the Appraisal Rights Notice in the form attached hereto as Exhibit K in accordance with Section 262 of the DGCL and a Letter of Transmittal to

- 9 -

each of the Stockholders other than the Requisite Stockholders who executed and delivered the Stockholder Consent. Any holder of Stock issued and outstanding immediately prior to the Effective Time with respect to which Appraisal Rights, if any, are available by reason of the Merger pursuant to Section 262 of the DGCL ("Dissenting Shares"), and have been perfected by such Stockholder, shall be entitled to receive only such rights as are granted under Section 262 of the DGCL and shall not be entitled to receive any portion of the Merger Consideration unless such Stockholder effectively withdraws or loses its Appraisal Rights under the DGCL. If any such holder effectively withdraws or loses such Appraisal Rights under the DGCL, such Dissenting Shares shall thereupon be deemed to have been converted as of the Effective Time into the right to receive that portion of the Merger Consideration due to such holder pursuant to this Agreement. Subject to the terms of Section 6.2, any payments made with respect to Dissenting Shares, or any costs incurred in connection with Dissenting Shares, that would be payable or reimbursable out of the Escrow Amount shall be made or incurred solely by the Surviving Corporation with the consent of the Equityholders' Representative, which shall not be unreasonably withheld.

2.8 Withholding Rights. The Purchaser (and after the Effective Time, the Surviving Corporation) shall be entitled to deduct and withhold from the Merger Consideration otherwise payable pursuant to this Agreement such amounts as it is required to deduct and withhold with respect to the making of such payment under the Code, or any provision of United States federal, state or local, or any foreign, Tax law. To the extent that amounts are so withheld or paid over to or deposited with the relevant Governmental Entity by the Purchaser (or Threewide), such amounts shall be treated for all purposes of this Agreement as having been paid to the applicable Equityholder in respect of which the Purchaser (or Threewide) made such deduction and withholding.

2.9 Certain Events at the Closing. At the Closing, in addition to such other actions as may be provided for herein:

(a) Purchaser shall deliver, to the Stockholders who have executed and delivered the Stockholder Consent and have executed and delivered a Letter of Transmittal and the Company Certificates in accordance with Section 2.5 and the Warrantholders who have executed and delivered a Letter of Transmittal and the Company Certificates in accordance with Section 2.5, an amount equal to such Equityholders' Pro Rata Share of the Merger Consideration (less such Equityholders' Pro Rata Share of the Holdbacks) in accordance with the provisions of this Agreement and the amounts set forth on Exhibit B;

(b) Purchaser shall deliver to the Escrow Agent an amount equal to the Escrow Amount for further payment in accordance with the provisions of this Agreement and the Escrow Agreement;

(c) Purchaser shall deliver to the Equityholders' Representative an amount equal to the Expenses Holdback for further payment in accordance with the provisions of this Agreement;

- 10 -

(d) Threewide shall pay the Management Carveout and Option Payment, subject to tax and other customary withholdings, to certain members of Threewide's management team as set forth on Exhibit F attached hereto;

(e) Threewide shall deliver Schedule 2.9(e) setting forth Threewide's Transaction Expenses;

(f) Purchaser shall deliver to Threewide the certificate required to be provided by it in Section 7.1(d);

(g) The Employment Agreements shall become effective;

(h) Purchaser, Threewide, the Equityholders' Representative and the Escrow Agent shall enter into the Escrow Agreement;

(i) Threewide shall deliver to Purchaser fully executed Option Termination Agreements; and

(j) The Parties shall file the Certificate of Merger.

3. REPRESENTATIONS AND WARRANTIES BY THREEWIDE

Threewide represents and warrants to the Purchaser and Merger Sub that the statements contained in this Section 3 and as set forth in the disclosure schedules delivered by Threewide to the Purchaser and Merger Sub on the date hereof (the "Disclosure Schedule") are correct and complete as of the date of this Agreement and will be correct and complete as of the Closing Date (as though made then and as though the Closing Date were substituted for the date of this Agreement throughout this Section 3). The Disclosure Schedule shall be arranged in paragraphs corresponding to the lettered and numbered paragraphs contained in this Section 3.

3.1 Capitalization.

(a) Threewide's authorized capital stock consists of the classes and shares of capital stock listed on Schedule 3.1(a) hereto, of which the number of shares of capital stock listed on Schedule 3.1(a) hereto are issued and outstanding. All outstanding shares of such capital stock are held of record and beneficially as set forth on Schedule 3.1(a) hereto.

(b) All of the issued and outstanding shares of the Stock have been duly and validly issued and are fully paid and nonassessable.

(c) Except as set forth on Schedule 3.1(c) hereto, there are no outstanding warrants, options, conversion privileges, preemptive rights, voting agreements or similar arrangements, or other rights or agreements to purchase or otherwise acquire or issue any equity securities of Threewide.

(d) Except as set forth on Schedule 3.1(d) hereto, there is no right of first refusal option or other restriction on transfer applicable to any of the shares of the Stock.

- 11 -

(e) Except as set forth on Schedule 3.1(e) hereto, Threewide does not have outstanding any stock appreciation rights or other rights granting to any Person the right to be paid money or other property based on the value of the securities of Threewide.

(f) Except as set forth on Schedule 3.1(f) hereto, there are no agreements, restrictions, or understanding to which Threewide is a party with respect to the sale, transfer, or voting of any shares of the Stock.

3.2 Organization and Good Standing.

(a) Threewide is: (i) a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware, with full corporate power and authority to conduct its business as it is now being conducted; and (ii) is duly qualified to operate its business, and is in good standing, in each jurisdiction where the ownership of property or the conducting of its business requires such qualification except where the failure to be so qualified would not have a Material Adverse Effect.

(b) Threewide has made available to Purchaser for its review and examination complete and accurate copies of Threewide's Certificate of Incorporation and Bylaws, each as amended through the date hereof, the minutes of all of Threewide's directors' and stockholders' meetings through the date hereof, and Threewide's stock transfer book, as updated through the date hereof, the Warrants, and all option agreements to acquire any shares of the Stock.

(c) Schedule 3.2(c) hereto lists all current officers and directors of Threewide as of the date hereof.

3.3 Subsidiaries. Threewide does not currently own, have any investment in, or control, directly or indirectly, any subsidiaries, associations or other business entities; and is not a participant in any joint venture or partnership.

3.4 Authority, Approval and Enforceability.

(a) Threewide has full corporate power and authority to execute, deliver and perform its obligations under this Agreement and, upon the execution and delivery of the Stockholder Consent by the Requisite Stockholders, all corporate action of Threewide necessary for such execution, delivery and performance has been duly taken. Complete and correct copies, certified by the Secretary or Assistant Secretary of Threewide, of the resolutions adopted by the Board of Directors, authorizing and ratifying the execution and delivery of this Agreement and the consummation of the transactions contemplated herein, have been delivered to the Purchaser.

(b) Except as provided in Schedule 3.4(b), the execution and delivery by Threewide of this Agreement and the agreements related hereto do not, and the performance and consummation of the transactions contemplated by this Agreement and the agreements related hereto will not, result in any conflict with, breach or violation of or default, termination or forfeiture under (or upon the giving of notice or the lapse of time, or both, result in any conflict with, breach or violation of or default, termination or forfeiture under) any terms or provisions of Threewide's Certificate of Incorporation or Bylaws, each as amended through the date hereof, or any statute, rule, regulation, judicial or governmental decree, order or judgment, or any of the

- 12 -

agreements set forth on Schedule 3.11 or Schedule 3.15(a), or any agreement related to any item set forth on Schedule 3.16(b) or Schedule 3.16(c).

(c) No consent, approval, authorization, order, registration, qualification or filing of or with any court or any regulatory authority or any other governmental body is required for the consummation by Purchaser and Threewide of the transactions contemplated by this Agreement.

(d) Upon due execution and delivery by the parties hereto, this Agreement and the agreements related hereto will each be legal, valid, and binding obligations of Threewide, enforceable against Threewide in accordance with their terms, except as may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' rights generally.

3.5 Financial Statements; Books of Account. An unaudited balance sheet of Threewide as of July 31, 2010 and the related statements of income and cash flows for the seven month period then ended and a scheduling of the liabilities, receivables, and fixed assets included thereon (collectively the "Threewide Financials") are included on Schedule 3.5 hereto. The Threewide Financials have been prepared in accordance with GAAP (except for the omission of certain footnotes) applied on a consistent basis throughout the periods involved. Except as set forth on Schedule 3.5 hereto, the books of account of Threewide reflect actual transactions as of the dates shown thereon and provide a true and correct representation of substantially all items of income and expenses, and substantially all assets, liabilities and accruals of Threewide. The Threewide Financials fairly present in all material respects the financial position, results of operations, and cash flows of Threewide covered thereunder as of the date thereof and for the periods covered thereby.

3.6 Material Changes. Except as set forth in Schedule 3.6, since July 31, 2010, there has not been with respect to Threewide any:

(a) Material Adverse Change in its financial condition from that shown on the Threewide Financials;

(b) Damage or loss, whether covered by insurance or not, materially and adversely affecting its business, property, assets or prospects;

(c) Other event or condition materially and adversely affecting its results of operations or business or financial condition or prospects taken as a whole or any event (not applicable to the economy or industry generally) which could have such an effect.

(d) Operation of the Business outside of the ordinary course;

(e) Violation of any law applicable to Threewide,

(f) Amendment, modification or supplement its organizational documents;

(g) Split, combination or reclassification of any of its capital stock or equity interests or issue or authorize the issuance of any securities in respect of, in lieu of, or in

- 13 -

substitution of its capital stock or equity interests or repurchase, redemption or other acquisition any of its capital stock or equity interests, except as expressly contemplated by this Agreement;

(h) Delay or postponement of payment of accounts payable or other liabilities or obligations or acceleration of the collection of accounts receivable, or write-downs of the value of any asset or write-offs of any accounts or notes receivable as uncollectible;

(i) Entrance into, termination of, amendment to or other modification or waiver of any of the terms of, any (i) agreement set forth on Schedule 3.15(a), (ii) employment agreement, (iii) real estate lease, or (iv) agreement between or among Threewide, on the one hand, and any Equityholder, director, or officer of Threewide or any Affiliate of any of the foregoing, on the other hand;

(j) (i) Adoption, amendment, modification or termination of any bonus, profit-sharing, incentive, severance or other benefit plan, contract or commitment for the benefit of any employee; (ii) increase of the compensation or other benefits payable or provided to any employee; or (iii) granting of any equity or equity based awards;

(k) Making or changing of any Tax election, changing of an annual accounting period, adopting or changing of any accounting method with respect to Taxes, filing of any amended Tax return, entrance into any closing agreement, settling or compromising of any proceeding with respect to any Tax claim or assessment relating to Threewide, surrendering of any right to claim a refund of Taxes, consenting to any extension or waiver of the limitation period applicable to any Tax claim or assessment relating to Threewide, or taking of any other similar action relating to the filing of any Tax return or the payment of any Tax;

(1) Declaration, setting aside, making or payment of any dividend or other distribution in respect of capital stock or equity interests of Threewide;

(m) Sale, assignment, license, pledge, transfer or disposition of, or granting or taking of any action that has created a lien on any assets, rights or properties, except for the sale of inventory to customers in the ordinary course of business consistent with past practice;

(n) Acquisition (by way of merger, consolidation or acquisition) of the equity interests or assets of any Person, or acquisition of direct or indirect control of any interest in any Person or the right to acquire any stock or interest in any other Person or in any business, or entrance into any joint venture, partnership or similar arrangement;

(o) (i) Distribution of any cash to any Equityholder or officer or director of Threewide except in the ordinary course of business consistent with past practice or (ii) making any commitment for any capital expenditures in excess of \$5,000 for any individual commitment or \$10,000 in the aggregate;

(p) Incurrence or guarantee of any indebtedness, issuance or sale of any debt securities or warrants or other rights to acquire debt securities, or entrance into any arrangement having the economic effect of any of the foregoing;

- 14 -

(q) Payment of or agreement to pay in settlement, or compromise or waiver of any rights under or pursuant to, any litigation, arbitration, investigation or other proceeding relating to Threewide;

(r) Change of the accounting methods or accounting practices followed by Threewide or change of the depreciation or amortization policies or rates;

(s) Cancellation or termination of any insurance policy or causing any of the coverage thereunder to lapse;

(t) Granting of any registration rights with respect to the equity interests of Threewide;

(u) Institution of any general layoff of employees or implementation of any early retirement plan or announcement of the planning of such a program; or

(v) Entrance into any agreement or committing to take any of the types of actions described in any of subsections (a) through (u) above.

3.7 Accounts Receivable.

(a) All accounts receivable reflected on the Threewide Financials are bona fide, arose from valid sales in the ordinary course of business in the aggregate amount thereof and, to Threewide's knowledge, are collectible in full, except to the extent of the reserve therefor on such financials. Schedule 3.7(a) contains a complete and accurate report showing all accounts receivable of Threewide outstanding as of the date of the Threewide Financials, together with an accurate aging of such accounts.

(b) None of Threewide's accounts receivable is subject to any lien or claim of offset, setoff or counterclaim and Threewide does not have any knowledge of any facts or circumstances that would give rise to any such lien or claim. Except as disclosed on Schedule 3.7(b), there are no accounts receivable which are contingent upon the performance by Threewide of future services.

3.8 No Undisclosed Liabilities. Threewide has no debts, liabilities or claims against it, contingent or otherwise, which would be of a nature required to be reflected in a balance sheet prepared in accordance with GAAP which are material individually or in the aggregate and which are not shown or fully provided for on the Threewide Financials, except debts, liabilities and claims incurred in the ordinary course of business since the date of the Threewide Financials which are not material in the aggregate. The applicable reserves reflected on the Threewide Financials are sufficient for payment of all claims, asserted and unasserted, of customers to which Threewide has sold products or provided services through such date. All products and services provided to customers by Threewide have complied in all material respects with all requirements binding upon Threewide, whether by law, regulation, agreement or otherwise.

- 15 -

3.9 Taxes.

(a) Except as set forth on Schedule 3.9(a), all federal, state, local and foreign Tax returns and reports required to be filed to date, and which are properly open for examination under applicable statutes of limitation, with respect to the operations of Threewide have been accurately prepared and duly filed, and all Taxes shown as payable on such returns and reports have been paid when due, including, without limitation income, withholding, payroll, sales and use, and real and personal property taxes. Except as set forth on Schedule 3.9(a) hereto: (i) Threewide has not executed or filed with any Taxing authority any agreement extending the period for assessment or collection of any Tax to a date subsequent to the date hereof; (ii) no issue has been raised by any federal, state, local or foreign Taxing authority in connection with an audit or examination of the Tax returns, business or properties of Threewide that has not been settled or resolved; (iii) there is no pending claim, asserted deficiency or assessment for additional Taxes which has not been paid, nor is there any basis for the assertion of any such claim, deficiency or assessment; (iv) no material special charges, penalties or fines have ever been asserted against Threewide with respect to payment of or failure to pay any Taxes; (v) the provision for Taxes shown on the Threewide Financials is sufficient for payment of all unpaid federal, state, local and foreign Taxes (whether asserted or unasserted) incurred by Threewide through such date.

(b) Threewide has not filed any consent to the application of Section 341(f) of the Internal Revenue Code of 1986, as amended (the "Code"), or been subject to any actual or deemed election under Section 338 of the Code.

3.10 Properties and Inventories.

(a) Threewide has good and marketable title to, valid leasehold interests in or other valid right to use all of the material assets used in its operations or necessary for the conduct of its business, free and clear of any material mortgages, pledges, security interests, licenses, encumbrances, restrictions or adverse claims, except for the lien of Taxes not yet due and payable. Schedule 3.10(a) contains a description and the location of any such material assets that are not in the possession of Threewide or that are located other than on Threewide's premises in Morgantown, West Virginia.

(b) All of Threewide's material assets are adequate and suitable for the purposes for which they are presently being used. Except for those items listed on Schedule 3.10(b) or assets subject to leases listed on Schedule 3.15(a), there are no items of equipment, machinery or other tangible assets of Threewide that are not currently being used in its business that are reflected on the Threewide Financials.

(c) There has not occurred, except for those items listed on Schedule 3.10(c), since the date of the Threewide Financials, any transfer of title other than in the ordinary course of business, any abandonment, any material pilferage or any material loss with respect to any material property or equipment of Threewide.

3.11 Real Property. Threewide has no interests in real property other than leasehold interests. Schedule 3.11 identifies all real property leased by Threewide. Threewide has

- 16 -

enforceable leasehold interests in all property shown on Schedule 3.11. The facilities of Threewide (i) do not encroach on the property of others, (ii) conform in all material respects with all applicable ordinances, regulations and zoning laws and (iii) are in good repair, reasonable wear and tear excepted. Threewide has not subleased, licensed, or otherwise granted any third party the right to use or occupy any real property leased by Threewide.

3.12 Environmental Matters. Threewide has complied in all material respects with any and all applicable statutes, rules and regulations in effect regarding the environment including, without limitation, statutes, rules and regulations regarding the production, handling, treatment and disposal of toxic chemicals and hazardous waste.

3.13 Health and Safety Matters. Threewide has complied in all material respects with any and all applicable health and safety statutes, rules and regulations of state, local and federal authorities in effect.

3.14 Insurance. Schedule 3.14 identifies all policies of insurance now in effect covering the assets, properties and business of Threewide and all products liability and life insurance policies maintained by Threewide. Threewide has made available a true and accurate copy of each of the policies listed on Schedule 3.14 to Purchaser. Threewide has not committed an act or omission which might invalidate any of such policies in whole or in part.

3.15 Purchase, Sale and Other Agreements.

(a) Except as set forth on Schedule 3.15(a), Threewide is not a party or subject to any oral or written:

i. agreement (or group of related agreements) for the purchase or sale of materials, supplies, products, or other personal property, or for the furnishing or receipt of services, the performance of which involve consideration in excess of \$25,000;

ii. agreement concerning a partnership or joint venture;

iii. agreement (or group of related agreements) under which it has created, incurred, assumed, or guaranteed any indebtedness for borrowed money, or any capitalized lease obligation or under which it has imposed an Encumbrance on any of its assets, tangible or intangible;

iv. agreement with any multiple listing service regarding access to any data of any such multiple listing service;

v. agreement with any third-party regarding distribution and/or syndication of the data of any multiple listing service, real estate broker, or real estate agent;

vi. material agreement concerning confidentiality or noncompetition;

vii. profit sharing, stock option, stock purchase, stock appreciation, deferred compensation, severance, or other material plan or arrangement for the benefit of its current or former directors, officers, and employees;

viii. collective bargaining agreement;

ix. agreement for the employment of any individual on a full-time, part-time, consulting, or other basis or providing severance benefits;

x. agreement under which it has advanced or loaned any amount to any of its directors, officers, and employees outside the ordinary course of business;

xi. agreement under which the consequences of a default or termination could have a Material Adverse Effect; or

xii. other agreement (or group of related agreements) the performance of which involves consideration in excess of \$25,000.

(b) A complete and accurate copy of each written agreement and other document identified on Schedule 3.15(a) has been made available to Purchaser. To Threewide's Knowledge, each agreement or arrangement identified on Schedule 3.15(a) is, except to the extent fully performed at the date hereof, in full force and effect and valid and binding in accordance with its terms in all material respects except as may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' rights generally and subject to general equity principles and to limitations on availability of equitable relief, including specific performance. Except as set forth on Schedule 3.15(b), to Threewide's Knowledge, no party to any such contract, agreement or arrangement is in material default under, or intends to cancel, withdraw, modify or amend, any such contract, agreement or arrangement.

(c) Except as set forth on Schedule 3.15(c), Threewide has performed all material obligations required to be performed by it on or prior to the date hereof under each material contract, obligation, commitment, agreement, undertaking, arrangement or lease referred to in this Agreement or in any exhibit hereto. Threewide has not received any notice that it is in default thereunder, nor, to Threewide's Knowledge, is there any basis for a default.

- 18 -

3.16 Trademarks, Patents and Other Intellectual Property Rights.

(a) Threewide has not interfered with, infringed upon, misappropriated, or violated any material Intellectual Property rights of third parties. Threewide has not received any charge, complaint, claim, demand, or notice alleging any such interference, infringement, misappropriation, or violation (including any claim that Threewide must license or refrain from using any Intellectual Property rights of any third party). To Threewide's Knowledge, no third party has interfered with, infringed upon, misappropriated, or violated any material Intellectual Property rights of Threewide.

(b) <u>Schedule 3.16(b)</u> identifies each patent or registration which has been issued to Threewide with respect to any of its Intellectual Property, identifies each pending patent application or application for registration which Threewide has made with respect to any of its Intellectual Property, and identifies each material license, agreement, or other permission which Threewide has granted to any third party with respect to any of its Intellectual Property (together with any exceptions). Threewide has made available to the Purchaser correct and complete copies of all such patents, registrations, applications, licenses, agreements, and permissions (as amended to date). <u>Schedule 3.16(b)</u> also identifies each material trade name or unregistered trademark used by Threewide in connection with any of its businesses. With respect to each item of Intellectual Property identified in <u>Schedule 3.16(b)</u>: (i) Threewide possesses all right, title, and interest in and to the item, free and clear of any encumbrance, license, or other restriction; (ii) the item is not subject to any outstanding injunction, judgment, order, decree, ruling, or charge; and (iii) no action, suit, proceeding, hearing, investigation, charge, complaint, claim, or demand is pending or, to Threewide's Knowledge, is threatened which challenges the legality, validity, enforceability, use, or ownership of the item.

(c) <u>Schedule 3.16(c)</u> identifies each material item of Intellectual Property that any third party owns and that Threewide uses pursuant to license, sublicense, agreement, or permission. Threewide has made available to the Purchaser correct and complete copies of all such licenses, sublicenses, agreements, and permissions (as amended to date). With respect to each such item of Intellectual Property required to be identified in <u>Schedule 3.16(c)</u>: (i) the license, sublicense, agreement, or permission covering the item is legal, valid, binding, enforceable, and in full force and effect in all material respects; (ii) no party to the license, sublicense, agreement, or permission is in material breach or default, and no event has occurred which with notice or lapse of time would constitute a material breach or default or permit termination, modification, or acceleration thereunder; (iii) no party to the license, sublicense, agreement, or permission has repudiated any material provision thereof; and (iv) Threewide has not granted any sublicense or similar right with respect to the license, sublicense, agreement, or permission.

(d) Threewide has taken all reasonable steps to maintain the confidentiality of and otherwise protect and enforce its rights in all Intellectual Property, including: (i) obtaining an appropriate non-disclosure agreement prior to disclosing any trade secrets to a third party; and (ii) imposing restrictions on unauthorized copying, unauthorized sale or transfer, recompilation, disassembly or reverse-engineering and other industry-standard restrictions on use prior to providing a third party with access to Intellectual Property.

- 19 -

3.17 Employees and Consultants.

(a) Schedule 3.17(a) identifies all currently effective consulting and employment agreements and other material agreements, either oral or written, with individual consultants or employees to which Threewide is a party. Complete and accurate copies of all such written agreements and summaries of all oral agreements have been made available to Purchaser. Also shown on Schedule 3.17(a) are the name of each officer, employee and agent of Threewide and each such person's present rate of regular compensation and bonus payments related to the fiscal year ending December 31, 2010. No current officer, manager or key employee of Threewide has notified Threewide of an intention to terminate employment or to seek a material change in his terms of employment. Except as set forth on Schedule 3.17(a) no employee of Threewide has accrued more than three weeks of paid vacation.

(b) Except as specifically disclosed on Schedule 3.17(b), Threewide is not a party to any pension, retirement, profit sharing, savings, bonus, incentive, deferred compensation, group health insurance or group life insurance plan or obligation, employee welfare benefit plan, or collective bargaining agreement or other agreement, written or oral, with any trade or labor union, employees' association or similar organization. With respect to each plan described on Schedule 3.17(b), Threewide has made available to Purchaser complete and accurate copies of the plan, the Internal Revenue Service determination letter, if any, all plan applications and amendments, the most recent plan actuarial reports and all reports of or regarding such plan required by the Employee Retirement Income Security Act of 1974, as amended, and any regulations issued thereunder ("ERISA"). With respect to each plan, if any, which is subject to ERISA, Threewide has properly prepared and timely filed all governmental reports and has properly and timely posted or distributed all notices and reports to employees required to be filed, posted or distributed with respect to such plan. No prohibited transaction has occurred with respect to any such plan which is subject to ERISA, nor is there any pending or, to Threewide's Knowledge, threatened assertion of the occurrence of any such transaction.

(c) Threewide has not received any documentation relating to union activities, including but not limited to, correspondence or orders from the National Labor Relations Board and any state labor relations agencies or organizations, and there are currently no agreements with any unions and no strikes or labor disputes pending or, to Threewide's Knowledge, threatened by any of the employees of Threewide.

(d) Threewide has complied in all material respects with all applicable laws or regulations relating to the employment of labor. Threewide has withheld all amounts required by law or agreement to be withheld from its employees for the payment of any Tax or contribution.

(e) There are no currently outstanding loans from Threewide to any officer, director or employee of Threewide and no commitments to lend any money or other property to any such person, other than, in each case, routine travel advances in the ordinary course of business.

(f) To Threewide's Knowledge, no employee is obligated under any agreement or judgment that would conflict with such employee's obligation to use his best efforts to promote the interests of Threewide or would conflict with Threewide's business as conducted

- 20 -

or proposed to be conducted. To Threewide's Knowledge, no employee of Threewide is in violation of the terms of any employment agreement or any other agreement relating to such employee's relationship with any previous employer and no litigation is pending or threatened with regard thereto.

3.18 Borrowings and Guarantees. Schedule 3.18 identifies all agreements and undertakings pursuant to which Threewide (i) is borrowing or is entitled to borrow any money, (ii) is lending or has committed itself to lend any money, or (iii) is a guarantor or surety with respect to the obligations of any Person. Complete and accurate copies of all such written agreements have been delivered to Purchaser.

3.19 Bank Accounts and Powers of Attorney. Schedule 3.19 identifies all bank accounts used in connection with the operations of the Business, whether or not such accounts are held in the name of Threewide, lists the respective signatories therefor, and lists the names of all Persons holding a power of attorney from Threewide.

3.20 Compliance with Laws. Except as set forth in Schedule 3.20, the present conduct of the Business by Threewide does not violate any law, ordinance, regulation, judgment, order, decree or rule of any court, arbitrator or governmental agency or entity in any respect material to the conduct of the Business, and there are no legal or administrative proceedings or investigations pending or, to Threewide's Knowledge, threatened, which, if determined adversely to Threewide, could reasonably be expected to result, individually or in the aggregate, in any Material Adverse Change.

3.21 Absence of Litigation. Except as set forth on Schedule 3.21, neither Threewide nor any officer or director of Threewide is engaged in, or has received any threat of, any litigation, arbitration, investigation or other proceeding relating to Threewide or its employee benefit plans, property, business assets, licenses, permits, Intellectual Property or goodwill, or against or affecting the transactions contemplated by this Agreement, nor, to the Knowledge of Threewide, is there any basis therefor.

3.22 No Brokers; Transaction Expenses. Neither Threewide nor any Equityholder is obligated for the payment of fees or expenses of any broker or finder in connection with the origin, negotiation or execution of this Agreement or in connection with any transaction contemplated hereby. Schedule 2.9(e) sets forth all of Threewide's Transaction Expenses.

3.23 Insider Transactions. Schedule 3.23 hereto is a true and complete list of all material contracts now in effect between Threewide and any Equityholder and a true and complete list of all material contracts and material transactions between Threewide and any Person who now is or at any time has been an officer, director or stockholder of Threewide, other than salary and incentive compensation arrangements of a customary nature entered into in the ordinary course of business. No director or officer of Threewide and, to the Knowledge of Threewide, no Affiliate of any of them has any interest (other than ownership of securities registered under the Securities Act of 1933, as amended) in excess of \$500 in (a) any equipment or other property, real or personal, tangible or intangible, including, but without limitation, any item of Intellectual Property, used in connection with or pertaining to the Business, or (b) any

- 21 -

creditor, supplier, customer, manufacturer, agent, representative or distributor of products of Threewide.

3.24 Accuracy of Documents and Information. The copies of all instruments, agreements, other documents and written information delivered to Purchaser by Threewide or any of its representatives are and will be true and correct copies as of the date of delivery thereof. No representations or warranties made by Threewide in this Agreement nor any document, written information statement, financial statement, certificate or exhibit prepared and furnished or to be prepared and furnished by Threewide, or its representatives, to Purchaser pursuant hereto or in connection with the transactions contemplated hereby, taken as a whole, contains or will contain any untrue statement of a material fact, or omits or will omit to state a material fact necessary to make the statements or facts contained herein or therein not misleading.

4. REPRESENTATIONS AND WARRANTIES BY PURCHASER

Purchaser represents and warrants to Threewide as follows:

4.1 Organization. Purchaser is a Delaware corporation, duly organized, legally existing and in good standing under the laws of the State of Delaware and has full power, ability and authority to conduct its business as it is now conducted, to enter into this Agreement and to carry out the other transactions and agreements contemplated hereby.

4.2 Due Authorization. The execution, delivery and performance of this Agreement and each of the other agreements contemplated hereby and the consummation of the transactions contemplated hereby and thereby have been duly authorized by all necessary company action of Purchaser. This Agreement has been duly executed and delivered by Purchaser and is a valid and binding obligation of Purchaser, fully enforceable in accordance with its terms. Neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby will: (i) conflict with or violate any provision of Purchaser's charter, bylaws, operating agreement or other governing instruments, or any law, ordinance or regulation or any decree or order of any court or administrative or other governmental body which is either applicable to, binding upon or enforceable against Purchaser; or (ii) result in any breach of or default under any material mortgage, contract, agreement, indenture, trust, written agreement or other instrument which is either binding upon or enforceable against Purchaser.

4.3 Litigation. There are no actions, suits, proceedings or investigations, either at law or in equity, or before any Governmental Entity in any United States or foreign jurisdiction, of any kind now pending or, to the best of the Purchaser's knowledge, threatened against the Purchaser that (i) question the validity of this Agreement or (ii) seek to delay, prohibit or restrict in any manner any action taken or to be taken by that Purchaser under this Agreement.

4.4 Brokers. All negotiations relative to this Agreement and the transactions contemplated by it have been carried on by the Purchaser directly with Threewide and without the intervention of any other Person and in a manner as not to give rise to any valid claim against any of the parties for any finder's fee, brokerage commission or like payment.

4.5 No Untrue Statements. No statement by Purchaser contained in this Agreement and no written statement contained in any certificate, schedule or other document required to be

- 22 -

furnished by Purchaser to Threewide pursuant to this Agreement contains or will contain any untrue statement of a material fact, or omits or will omit to state a material fact necessary in order to make the statements not misleading.

4.6 Merger Sub. The representations set forth in Sections 4.1-4.5 and true and correct with respect to Merger Sub as if "Merger Sub" were substituted for "Purchaser" each place it appears.

5. COVENANTS

5.1 Further Cooperation. In case at any time after the Closing Date any further actions are necessary to carry out the purposes of this Agreement, each of the Parties will take such further actions (including the execution and delivery of such further instruments and documents) as the other Party reasonably may request, all at the sole cost and expense of the requesting Party (unless the requesting Party is entitled to indemnification therefor under Section 6 below).

5.2 Employees. Subject to the requirements, if any, of applicable law and obtaining approvals, if any, of Governmental Entities, prior to the Closing Date, Threewide shall take all actions necessary to provide for the cancellation or termination, effective immediately prior to the Closing, of any Threewide 401(k) or profit sharing plan. Effective as of the Closing Date, Buyer shall offer to all employees of Threewide as of the Closing Date ("Continuing Employees") compensation and benefits that are substantially comparable to the compensation and benefits they were receiving from Threewide immediately prior to the Closing Date. Comparable benefits shall be offered to Continuing Employees either through the Buyer's continuing to maintain the existing Threewide Employee Welfare Benefit Plans or through immediate participation in Employee Welfare Benefit Plans maintained by Purchaser for its similarly-situated employees. Continuing Employees shall be eligible to participate in any Employee Pension Benefit Plan that Purchaser maintains for its similarly-situated employees effective no later than the first day of the first full month immediately following the Closing Date.

5.3 Stockholder Consent and Appraisal Rights. Threewide agrees to carry out the provisions of Sections 2.5 and 2.7 hereof and to comply with Threewide's certificate of incorporation, bylaws and the applicable provisions of the DGCL (including Section 262 of the DGCL).

5.4 Conduct of Business Prior to Closing. Threewide agrees that, from the date hereof until the Closing, the business of Threewide shall be conducted in the ordinary course of business consistent with past practice.

6. SURVIVAL, INDEMNIFICATION

6.1 Survival. All representations, warranties and covenants made and given in this Agreement (including any exhibit hereto), including those contained in any Schedule or any certificate delivered on the Closing Date by any party shall be deemed to have been relied on in spite of any investigation that has been or will be made or omitted by any party to this Agreement and shall survive for 2 years following the Closing Date.

- 23 -

6.2 Indemnification.

(a) To the limited extent set forth in this Section 6.2, the Equityholders shall indemnify, defend and save Purchaser and its Affiliates (including Threewide from and after the Closing) and the stockholders, directors, officers, employees, and agents of Purchaser and its Affiliates (including Threewide from and after the Closing) and the stockholders, directors, officers, employees, and agents of Purchaser and its Affiliates (including Threewide from and after the Closing) and the successors and assigns of the foregoing (collectively, the "Purchaser Indemnified Parties") harmless from any actions, claims, losses, damages, demands or expense (including without limitation all court costs and reasonable attorney's fees on account thereof) suffered or incurred by any of them proximately caused by (i) any inaccuracy in, nonperformance of, or breach of any representation, warranty, covenant, obligation or other undertaking made by Threewide set forth in this Agreement (including any exhibit hereto) and (ii) any Appraisal Rights.

(b) No indemnification shall be payable to the Purchaser Indemnified Parties under this Section 6.2 until and after the Purchaser Indemnified Parties' losses aggregate \$25,000 (the "Basket"); provided, however, that in the event that such losses exceed the amount of the Basket, then indemnification shall be made by the Equityholders for the full amount of such losses, including the initial \$25,000; provided, further, that the Basket shall not apply to (i) any actions, claims, losses, damages, demands or expense (including without limitation all court costs and reasonable attorney's fees on account thereof) suffered or incurred by the Purchaser Indemnified Parties proximately caused by any inaccuracy in, nonperformance of, or breach of any representation or warranty set forth in Section 3.1 (Capitalization) or in sub-sections (a), (c), or (d) of Section 3.4 (Authority, Approval and Enforceability), (ii) the indemnification obligations pursuant to Section 6.2(a)(ii), or (iii) the indemnification obligations pursuant to Section 6.2(d). Purchaser shall notify the Equityholders' Representative promptly of any written actions, claims or demands against any Purchaser Indemnified Parties of which the Equityholders are responsible hereunder, specifying the basis and amount thereof in reasonable detail.

(c) Notwithstanding any provision to the contrary contained elsewhere in this Agreement, the Equityholders' combined aggregate liability for indemnification under Section 6.2(a) hereof shall not exceed the amount then held in the Escrow Account (the "Cap") and no claim shall be made directly against any Equityholder; provided, however, that the Cap shall not apply to any actions, claims, losses, damages, demands or expenses as a result of breaches or other inaccuracies in the Letters of Transmittal and the Purchaser Indemnified Parties may make direct claims against Equityholders for such breaches or other inaccuracies.

(d) The Equityholders shall indemnify, defend and save the Purchaser Indemnified Parties harmless from any Transaction Expenses of Threewide exceeding the Expenses Cap, including any actions, claims, losses, damages, demands or expense (including without limitation all court costs and reasonable attorney's fees on account thereof) related thereto.

(e) Purchaser shall indemnify, defend and save Threewide and the Equityholders harmless from any actions, claims, losses, damages, demands or expense (including without limitation all court costs and reasonable attorney's fees on account thereof) suffered or incurred by Threewide or the Equityholders, or their successors and assigns,

- 24 -

proximately caused by any breach of any representation, warranty, covenant or other undertaking made by the Purchaser in this Agreement (including any exhibit hereto). Threewide and/or the Equityholders' Representative shall notify the Purchaser promptly of any written actions, claims or demands against Threewide or the Equityholders of which the Purchaser is responsible hereunder specifying the basis and amount thereof in reasonable detail.

6.3 Indemnification Procedures.

(a) Promptly after any Party or other Person entitled to indemnification under this Section 6 (each, an "Indemnified Party") determines that a claim for indemnification might exist, including notice of any claim by a third party described in Section 6.3(c), the Indemnified Party shall promptly deliver a certificate (a "Certificate") to the Indemnifying Party. The Person or Persons who are required to indemnify such Indemnified Party under this Section 6 shall be referred to herein as the "Indemnifying Party." Such Certificate shall:

i. state that the Indemnified Party may have incurred, or reasonably anticipates that it will incur, damages for which such Indemnified Party is entitled to indemnification pursuant to this Agreement; and

ii. specify such damages in reasonable detail included in the amount so stated, the date such item was discovered, the basis for any anticipated damages and the nature of the misrepresentation, inaccuracy or claim to which each such item is related and the computation of the amount to which such Indemnified Party claims to be entitled under Section 6.2 of this Agreement. Notwithstanding any other provision of this Agreement to the contrary, for purposes of Section 6.2(a), with respect to each representation or warranty contained in this Agreement that is qualified by knowledge, "to the Knowledge of", materiality, "Material Adverse Effect," or other similar words or phrases, no such qualification shall be permitted for the purpose of determining the amount of any actions, claims, losses, damages, demands or expenses that are the subject of indemnification hereunder, and all such actions, claims, losses, damages, demands and expenses shall be calculated without regard to any such qualification.

(b) In case the Indemnifying Party shall object to the indemnification of an Indemnified Party in respect of any claim or claims specified in any Certificate, the Indemnifying Party shall, within 30 days after receipt by the Indemnifying Party of such Certificate, deliver to the Indemnified Party a written notice to such effect and the Indemnifying Party and the Indemnified Party shall, within the 30 day period beginning on the date of receipt by the Indemnified Party of such written objection, attempt in good faith to agree upon the rights of the respective parties with respect to each of such claims to which the Indemnifying Party shall have so objected. If the Indemnified Party and the Indemnifying Party shall succeed in reaching agreement on their respective rights with respect to any of such claims, the Indemnified Party and the Indemnifying Party shall promptly prepare and sign a memorandum setting forth such agreement. Should the Indemnified Party and the Indemnifying Party be unable to agree as to any particular item or items or amount or amounts, then such dispute shall be settled by arbitration in Wilmington, Delaware, in accordance with the Commercial Arbitration Rules of the American Arbitration Association then in effect. There shall be three arbitrators, one to be chosen by each of the Indemnified Party and Indemnifying Party directly at will, and the third arbitrator to be selected by the two arbitrators so chosen. Each arbitrator shall be an attorney: (i)

- 25 -

whose primary practice area comprises mergers and acquisitions; (ii) with at least 10 years of practice experience; and (iii) who is a partner of a law firm. Each of the Indemnifying Party and the Indemnified Party shall pay the fees of the arbitrator he or it selects and of his or its own attorneys and the expenses of his or its witnesses, and all other fees and costs shall be borne equally by the Indemnifying Party and the Indemnified Party. Judgment on any award rendered by the arbitrators may be entered in any court having jurisdiction and neither the Indemnifying Party nor the Indemnified Party shall object to the entry of such award.

(c) Third Party Claims.

i. If any third party notifies any Party (the "Indemnified Party") with respect to any matter (a "Third-Party Claim") that may give rise to a claim for indemnification against the other Party (the "Indemnifying Party") under this Section 6, then the Indemnified Party shall promptly (and in any event within five business days after receiving notice of the Third-Party Claim) notify the Indemnifying Party thereof in writing specifying the claimed basis and amount thereof in reasonable detail. Failure to so notify shall not be deemed a waiver of, or otherwise affect, the Indemnifying Party's obligations unless the Indemnifying Party is harmed or prejudiced by such failure to notify.

ii. The Indemnifying Party will have the right at any time to assume and thereafter conduct the defense of the Third-Party Claim with counsel of its choice; provided, however, that the Indemnifying Party will not consent to the entry of any judgment on or enter into any settlement with respect to the Third-Party Claim without the prior written consent of the Indemnified Party (not to be unreasonably withheld) unless the judgment or proposed settlement involves only the payment of money damages and does not impose an injunction or other equitable relief upon the Indemnified Party.

iii. Unless and until the Indemnifying Party assumes the defense of the Third-Party Claim as provided in above, however, the Indemnified Party may defend against the Third-Party Claim in any manner it may reasonably deem appropriate. In no event will the Indemnified Party consent to the entry of any judgment on or enter into any settlement with respect to the Third-Party Claim without the prior written consent of the Indemnifying Party (not to be unreasonably withheld).

(d) Claims specified in any Certificate to which an Indemnifying Party shall not object in writing within 30 days of receipt of such Certificate or that have been resolved pursuant to Section 6.3(c) are hereinafter referred to as "Agreed Claims". Within 30 days after the determination of the amount of any Agreed Claims, the Indemnifying Party shall pay to the Indemnified Party an amount equal to the Agreed Claim by wire transfer in immediately available funds to the bank account or accounts designated in writing by the Indemnified Party not less than five days prior to such payment, provided that any Agreed Claim to be paid by the Equityholders shall be payable solely through the Escrow Account.

- 26 -

7. CONDITIONS TO CLOSING; FAILURE TO CLOSE.

7.1 Conditions to Purchaser's Obligations at Closing. All of the obligations of the Purchaser under this Agreement are subject to the fulfillment at or before the Closing of each of the following conditions, any of which may be waived in writing by Purchaser:

(a) Representations and Warranties. The representations and warranties of Threewide contained herein shall be true and correct in all material respects (without giving effect to the knowledge or materiality qualifiers contained therein) on and as of the Closing Date with the same effect as if made on and as of the Closing Date and Purchaser shall have received a copy of the Disclosure Schedules updated through and dated as of the Closing Date.

(b) Performance. Threewide shall have performed or fulfilled in all material respects all agreements, obligations and conditions contained herein and shall have obtained all consents, waivers and approvals set forth on Schedule 7.1(b), including those necessary to consummate the transactions contemplated hereby, including the Stockholder Consent executed by the Requisite Stockholders.

(c) Material Adverse Change. There shall have been no Material Adverse Change.

(d) Certificate. Purchaser shall have received at the Closing a certificate, dated as of the Closing and executed by Threewide's President, to the effect that the conditions set forth in Sections 7.1(a) - (c) have been satisfied.

(e) Related Documents. Each of the following agreements shall have been duly executed and delivered by the parties thereto (other than the Purchaser and Merger Sub): (i) the Certificate of Merger; (ii) the Escrow Agreement; and (iii) the Option Termination Agreements.

(f) Resignation Letters. Threewide shall have delivered to the Purchaser the resignation letters of all members of the board of directors of Threewide and/or any officer of Threewide as the Purchaser shall have requested at or prior to the Closing.

(g) No Order. No action, suit, proceeding or investigation shall have been adopted or promulgated and no Governmental Entity shall have issued any order that is in effect, which has the effect of making illegal or otherwise prohibiting the consummation of the transactions contemplated by this Agreement.

(h) Legal Opinion. Purchaser shall have received an executed copy of a favorable written opinion of Thorp Reed & Armstrong, LLP, counsel for Threewide, in form and substance reasonably satisfactory to Purchaser.

- 27 -

7.2 Conditions to Threewide's Obligations at Closing. The obligations of Threewide under this Agreement are subject to the fulfillment at or before the Closing of each of the following conditions, any of which may be waived in writing by Threewide:

(a) Representations and Warranties. The representations and warranties of the Purchaser contained herein shall be true and correct in all material respects (without giving effect to the knowledge or materiality qualifiers contained therein) on and as of the Closing Date with the same effect as if made on and as of the Closing Date.

(b) Performance. The Purchaser shall have performed or fulfilled in all material respects all agreements, obligations and conditions contained herein and shall have obtained all consents, waivers and approvals set forth on Schedule 7.2(b), including those necessary to consummate the transactions contemplated hereby, including the Stockholder Consent executed by the Requisite Stockholders.

(c) Aggregate Merger Consideration. At the Closing, Purchaser shall deliver the Merger Consideration in accordance with Section 2.9(a) via wire transfer in immediately available funds, and the Escrow Agent shall confirm receipt via wire transfer in immediately available funds of the Escrow Amount.

(d) Related Documents. Each of the following agreements (such agreements, together with this Agreement, the "<u>Related Documents</u>") shall have been duly executed and delivered by the parties thereto (other than Threewide): (i) the Certificate of Merger; (ii) the Escrow Agreement; and (iii) the Option Termination Agreements.

7.3 Failure to Perform by Threewide. If the conditions to Closing set forth in Section 7.2 shall be met but Threewide refuses to consummate the Merger, then Purchaser shall have the option to <u>either</u> (a) obtain specific performance of Threewide to consummate the Merger, upon which Threewide hereby consents to agree to so perform; or (b) recover from Threewide, as liquidated damages, the sum of \$2,000,000, which amount represents a good-faith estimate of the amount of damages that will likely be sustained by Purchaser as the result of a breach of this Agreement by Threewide, it being impracticable or extremely difficult to fix the actual damages.

8. TERMINATION AND EQUITYHOLDERS REPRESENTATIVE

8.1 Events of Termination. This Agreement may be terminated: (i) at any time by mutual written agreement of the Parties; (ii) in whole and not in part by the Purchaser by written notice to Threewide if the conditions set forth in Section 7.1 hereof shall not have been complied with or performed on or prior to September 30, 2010 (or such later date as the Parties may have agreed to in writing) in any material respect and the Purchaser shall not have materially breached any of its representations, warranties, covenants or agreements contained herein; (iii) in whole and not in part by Threewide by written notice from Threewide to the Purchaser if the conditions set forth in Section 7.2 hereof shall not have been complied with or performed on or prior to September 30, 2010 (or such later date as the Parties may have agreed to in writing) in any material respect and Threewide shall not have materially breached any of its representations, warranties, covenants or agreements or agreements contained herein; or (iv) by any Party, by written notice to the other Parties, if a court of competent jurisdiction or other Governmental Entity shall have

- 28 -

issued a final, non-appealable order, decree or ruling, or taken any other action, having the effect of permanently restraining, enjoining or otherwise prohibiting the transactions contemplated by this Agreement.

8.2 Effect of Termination. In the event that this Agreement shall be terminated pursuant to Section 8.1, all further obligations of the Parties hereto under this Agreement (other than pursuant to Sections 7.3(b), 9.1 through 9.14, which shall continue in full force and effect) shall terminate without further liability or obligation of any Party to any other Party hereunder; provided, however, that no Party shall be released from liability hereunder if this Agreement is terminated and the transactions abandoned by reason or failure of such Party to have performed its obligations hereunder. In addition, in the event of a termination of this Agreement, Purchaser shall keep confidential any information obtained from Threewide concerning its properties, operations and business (unless readily ascertainable from public or published information or trade sources) until the same ceases to be material (or becomes so ascertainable) and, at the request of Threewide shall return to Threewide all copies of any schedules, statements, documents or other written information obtained in connection therewith. Without limiting the generality of the foregoing, the terms of the Mutual Confidentiality and Non-Disclosure Agreement dated January 19, 2010 between Move, Inc. and Threewide Corporation relating to obligations with respect to confidential information and nonsolicitation are hereby incorporated by reference and, notwithstanding the termination of this Agreement, such terms shall continue in full force and effect until the earlier of the Closing and the date that such Mutual Confidentiality and Non-Disclosure Agreement is terminated and effect until the earlier of the

8.3 Equityholders' Representative.

(a) The Equityholders' Representative is hereby designated to act following the Closing with respect to any and all issues arising under this Agreement and, to the extent set forth herein, the Equityholders' Representative shall take, and is hereby authorized to take, any and all actions that he believes are necessary or appropriate under this Agreement as permitted above, as fully as if the Equityholders were acting on their own behalf and with the same force and effect, including, without limitation, by:

i. Consenting to, compromising or settling any and all claims arising under this Agreement;

ii. Conducting negotiations with the Purchaser and its representatives regarding such claims;

iii. Performing all of his obligations under this Agreement;

iv. Waiving, modifying, amending or supplementing provisions of this Agreement and the obligations of the Purchaser hereunder;

v. Taking any and all other actions specified in or contemplated by this Agreement; and

vi. Engaging counsel, accountants or other representatives in connection with the foregoing matters. The Purchaser shall have the right to rely

upon all actions taken or omitted to be taken by either one of the individuals acting as the Equityholders' Representative in respect of this Agreement and the transactions contemplated hereby, all of which actions or omissions shall be legally binding upon the Equityholders, and the Purchaser shall not be liable for any action taken or not taken in good faith reliance on a communication or other instruction from the Equityholders' Representative.

(b) Under no circumstances shall the Equityholders' Representative have any liability to any Equityholder for any act or omission to act of the Equityholders' Representative in such capacity, unless the party asserting such liability is able to prove that the Equityholders' Representative was guilty of gross negligence or willful misconduct. The Equityholders' Representative shall be entitled to payment of or reimbursement for any and all costs and expenses incurred in connection with, arising out of, resulting from or incident to any act or omission to act of the Equityholders' Representative in his capacity as such, except to the extent caused by the gross negligence or willful misconduct of the Equityholders' Representative. Such costs and expenses shall be paid solely out of the Expense Holdback as contemplated herein.

(c) If at any time the Equityholders' Representative or his successor is unwilling or unable to serve as the Equityholders' Representative, a new Equityholders' Representative shall be promptly designated. The new Equityholders' Representative shall be designated by the approval of Stockholders who collectively held at least a majority of Threewide's Series C Preferred Stock outstanding immediately prior to the Closing.

9. GENERAL PROVISIONS

9.1 No Waivers. None of the Parties shall be deemed to waive any of its rights, powers or remedies hereunder unless such waiver is in writing and signed by said Party. No delay or omission by any Party in exercising any of said rights, powers or remedies shall operate as a waiver thereof, nor shall a waiver signed by any Party of any breach of the covenants, conditions or agreements binding on the other Parties on one occasion be construed as a waiver or consent to such breach on any future occasion or a waiver of any other covenant, condition, or agreement herein contained.

9.2 Expenses. Purchaser and Merger Sub shall bear their Transaction Expenses, and the Transaction Expenses of Threewide set forth on Schedule 2.9(e) shall increase the Merger Consideration as provided in Section 2.3(a)(iii). The Equityholders shall bear the Transaction Expenses of Threewide exceeding the Expenses Cap in accordance with Section 6.2(d).

9.3 Publicity. Each Party shall obtain the written consent of the other Parties prior to any publication, presentation, public announcement or press release concerning the relationship between the Parties or the existence or terms of this Agreement, except as may otherwise be required by law. In addition, the Parties agree not to make any disparaging or derogatory comments regarding the other Parties to any third party.

9.4 Assignment. No Party may assign any portion of this Agreement, voluntarily or involuntarily, including without limitation by operation of law, without the prior written consent of the other Parties. Any attempt to otherwise assign this Agreement shall be null and void.

- 30 -

Notwithstanding the foregoing, Purchaser shall be permitted to assign this Agreement, including without limitation by operation of law, to any Affiliate of Purchaser and any of Purchaser's or its Affiliate's lenders without the prior written consent of any other Party. No Person not a Party hereto, other than the Equityholders, shall have any interest herein or be deemed a third party beneficiary hereof, and nothing contained herein shall be construed to create any rights enforceable by any other Person or third party; provided, however, that Purchaser or its Affiliates may pledge or collaterally assign this Agreement and its or their rights hereunder to any of Purchaser's or its Affiliate's lenders.

9.5 Partnership. Nothing herein contained shall be construed as creating a partnership or joint venture by or between the Parties.

9.6 Binding Agreement. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective permitted successors and assigns.

9.7 Severability. Any provision of this Agreement held or determined by a court (or other legal authority) of competent jurisdiction to be illegal, invalid, or unenforceable in any jurisdiction shall be deemed separate, distinct and independent, and shall be ineffective to the extent of such holding or determination without (i) invalidating the remaining provisions of this Agreement in that jurisdiction or (ii) affecting the legality, validity or enforceability of such provision in any other jurisdiction.

9.8 Captions Headings. Captions and paragraph headings used in this Agreement are for convenience only and shall not be used to interpret any provision hereof.

9.9 Entire Agreement. This Agreement, together with the Exhibits and Disclosure Schedules, which documents are incorporated herein by reference, constitutes the entire agreement and understanding of the parties with respect to the subject matter hereof, and is intended as the parties' final expression and complete and exclusive statement of the terms thereof, superseding all prior or contemporaneous agreements, representations, promises and understandings, whether written or oral, between the Parties, and may be amended or modified only by an instrument in writing signed by all Parties.

9.10 Notices. Any notice required or permitted to be given hereunder shall be (a) in writing, (b) effective on the first business day following the date of receipt, and (c) delivered by one of the following means: (i) by personal delivery; (ii) by prepaid, overnight package delivery or courier service; or (iii) by the United States Postal Service, first class, certified mail, return receipt requested, postage prepaid. All notices given under this Agreement shall be addressed:

in the case of the Equityholders' Representative, to:

William Rice 1411 Connell Road Charleston, WV 25314 and

J. Rudy Henley Mountaineer Capital LLC 107 Capitol Street

- 31 -

Suite 300 Charleston, WV 25301

with a copy to:

John C. Rodney, Esq. Thorp Reed & Armstrong, LLP One Oxford Centre 301 Grant Street Pittsburgh, Pennsylvania 15219-1425

in the case of Threewide, to:

Threewide Corporation 709 Beechurst Avenue, Suite 1B Morgantown, West Virginia 26505 Attention: Luke Glass

with a copy to:

John C. Rodney, Esq. Thorp Reed & Armstrong, LLP One Oxford Centre 301 Grant Street Pittsburgh, Pennsylvania 15219-1425

All notices given under this Agreement shall be addressed, in the case of Purchaser or Merger Sub, as follows:

Move Sales, Inc. c/o Move, Inc. Attn: General Counsel 30700 Russell Ranch Road Westlake Village, California 91362

or to such other addresses of which the Parties have been advised in writing by any of the above-described means. Personal delivery to a Party or to any officer, partner, agent, or employee of such Party at its address herein shall constitute receipt. The following shall also constitute receipt: (i) a Party's rejection or other refusal to accept notice, and (ii) the inability to deliver to a Party because of a changed address of which no notice has been received by the other Party. Notwithstanding the foregoing, no notice of change of address shall be effective until 10 days after the date of receipt thereof. This Section shall not be construed in any way to affect or impair any waiver of notice or demand herein provided.

9.11 Governing Law. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware, without giving effect to any principles of conflicts of law. With respect to any litigation arising out of or relating to this Agreement, the

- 32 -

Parties agree that it shall be exclusively filed in and heard by the state or federal courts with jurisdiction to hear such suits located in the State of Delaware, and each Party hereby submits to the exclusive jurisdiction of such courts.

9.12 Counterparts. This Agreement may be executed in multiple counterparts, each of which shall be deemed an original and all of which taken together shall constitute one and the same Agreement. Execution and delivery of this Agreement by exchange of facsimile copies bearing the facsimile signature of a Party hereto shall constitute a valid and binding execution and delivery of this Agreement by such Party. Such facsimile copies shall constitute enforceable original documents.

9.13 Specific Performance. Without limiting or waiving in any respect any rights or remedies of Purchaser now or hereinafter existing at law or equity or by statute, Purchaser shall be entitled to seek specific performance of the obligations to be performed by the other Parties in accordance with the provisions of this Agreement.

9.14 Professional Fees. In the event of any litigation involving this Agreement or any of the ancillary documents, the prevailing Party in such litigation shall be entitled to recover reasonable attorney fees and costs in addition to any other remedy to which it is entitled.

[SIGNATURES ON FOLLOWING PAGE]

- 33 -

IN WITNESS WHEREOF, each Party has executed or caused its duly authorized officer to execute this Agreement the day and year first above written.

PURCHASER:

Move Sales, Inc.

By: /s/ Errol Samuelson

Name: Errol Samuelson Its: Chief Revenue Officer

THREEWIDE:

Threewide Corporation

By: /s/ Luke Glass

Name: Luke Glass Its: President

MERGER SUB:

MSI TC Merger Sub, Inc.

By: /s/ Errol Samuelson Name: Errol Samuelson Its: Chief Revenue Officer

EQUITYHOLDERS' REPRESENTATIVE:

/s/ William Rice William Rice

/s/ J. Rudy Henley J. Rudy Henley

[Signatures Continue on Following Page]

The undersigned executes this Agreement for the purpose of guaranteeing the obligations of Move Sales, Inc. hereunder.

Move, Inc.

By: <u>/s/ Errol Samuelson</u> Name: Errol Samuelson Its: Chief Revenue Officer

[Signatures Continue on Following Page]

The undersigned Requisite Stockholders execute this Agreement for the purpose of agreement to execute the Stockholder Consent immediately after the execution of this Agreement and to otherwise vote their shares of the Stock to cause the approval of the transactions contemplated hereby. The undersigned Requisite Stockholders hereby represent and warrant to Purchaser that as of the date hereof there are no outstanding voting proxies, voting trusts or voting agreements with respect to their shares of the Stock and the undersigned agree not to grant any such voting proxy or enter into any such voting trust or agreement subsequent to the date hereof.

Mountaineer Capital, L.P.

By: /s/ Patrick A. Bond Name: Patrick A. Bond Its: General Partner

Adena Ventures, L.P.

By: <u>/s/ Lynn Gellermann</u> Name: Lynn Gellermann Its: President

Select Capital Ventures I, L.P.

By: /s/ Rocco A. Ortenzio

Name: Rocco A. Ortenzio Its: Managing Member

West Virginia Jobs Investment Trust Board

By: /s/ C. Andrew Zulauf Name: C. Andrew Zulauf Its: Ex. Director

/s/ William Rice William Rice

/s/ William McDavid William McDavid

[Signature Page to Agreement and Plan of Merger]

The undersigned Warrantholders execute this Agreement for the purpose of agreeing to the terms of this Agreement relating to the Warrants, including the treatment thereof, notwithstanding the terms and conditions set forth in the Warrants. Each of the undersigned Warrantholders acknowledge and agree that the Warrants held by each such Warrantholder as of the date of this Agreement are set forth opposite each such Warrantholders' name on Exhibit E attached hereto. Each of the undersigned Warrantholders acknowledges and agrees that, notwithstanding anything to the contrary set forth in the Warrants, pursuant to the terms and conditions of this Agreement, (a) at the Closing, the Warrants shall be converted into the right to receive a portion of the Merger Consideration (less the amount of the Holdbacks) as set forth on Exhibit B hereto, (b) the Holdbacks will be applied as set forth in Section 2.4(a), and (c) once the Warrants are so converted, such Warrants shall no longer be outstanding and shall terminate in all respects. Each of the undersigned Warrantholders covenants not to exercise or transfer or assign any of the Warrants held by such Warrantholder following the date of this Agreement. Each of the undersigned Warrantholders hereby waives any right to prior notice of any events contemplated by this Agreement.

Mountaineer Capital, L.P.

By: /s/ Patrick A. Bond	/s/ William Rice		
Name: Patrick A. Bond	William Rice		
Its: General Partner			
Adena Ventures, L.P.	/s/ William McDavid		
	William McDavid		
By: /s/ Lynn Gellermann			
Name: Lynn Gellermann			
Its: President	/s/ Kenneth Kleeman		
	Kenneth Kleeman		
Select Capital Ventures I, L.P.			
	/s/ Charles Smith		
By: : /s/ Rocco A. Ortenzio	Charles Smith		
Name: : Rocco A. Ortenzio			
Its: Managing Member			
West Virginia Jobs Investment Trust Board			

By: <u>/s/ C. Andrew Zulauf</u> Name: C. Andrew Zulauf Its: Ex. Director

[Signature Page to Agreement and Plan of Merger]

CERTIFICATION

I, Steven H. Berkowitz, certify that:

1. I have reviewed this quarterly report on Form 10-Q of Move, Inc.;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

(a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

(b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

(c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

(d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ STEVEN H. BERKOWITZ Steven H. Berkowitz *Chief Executive Officer*

Date: November 5, 2010

CERTIFICATION

I, Robert J. Krolik, certify that:

1. I have reviewed this quarterly report on Form 10-Q of Move, Inc.;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

(a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

(b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

(c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

(d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ ROBERT J. KROLIK Robert J. Krolik Chief Financial Officer

Date: November 5, 2010

STATEMENT OF CHIEF EXECUTIVE OFFICER OF MOVE, INC. PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO § 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the report of Move, Inc. (the "Company") on Form 10-Q for the period ended September 30, 2010 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), the undersigned Steven H. Berkowitz, Chief Executive Officer of the Company, certifies, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that:

1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Move, Inc. Date: November 5, 2010 /s/ STEVEN H. BERKOWITZ

Steven H. Berkowitz Chief Executive Officer

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to Move, Inc. and will be retained by Move, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.

STATEMENT OF CHIEF FINANCIAL OFFICER OF MOVE, INC. PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO § 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the report of Move, Inc. (the "Company") on Form 10-Q for the period ended September 30, 2010 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), the undersigned Robert J. Krolik, Chief Financial Officer of the Company, certifies, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that:

1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Move, Inc. Date: November 5, 2010 /s/ ROBERT J. KROLIK Robert J. Krolik Chief Financial Officer

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to Move, Inc. and will be retained by Move, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.